

Temi di discussione

(Working Papers)

Energy price shocks, unemployment, and monetary policy

by Nicolò Gnocato





Temi di discussione

(Working Papers)

Energy price shocks, unemployment, and monetary policy

by Nicolò Gnocato

Number 1450 - March 2024

The papers published in the Temi di discussione series describe preliminary results and are made available to the public to encourage discussion and elicit comments.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Antonio Di Cesare, Raffaela Giordano, Marco Bottone, Lorenzo Braccini, Mario Cannella, Alessandro Cantelmo, Giacomo Caracciolo, Antoniomaria Conti, Antonio Dalla Zuanna, Valerio Della Corte, Marco Flaccadoro, Rosalia Greco, Alessandro Moro, Stefano Piermattei, Fabio Piersanti, Dario Ruzzi. *Editorial Assistants:* Roberto Marano, Marco Palumbo, Gwyneth Schaefer.

ISSN 2281-3950 (online)

Designed by the Printing and Publishing Division of the Bank of Italy

ENERGY PRICE SHOCKS, UNEMPLOYMENT, AND MONETARY POLICY

by Nicolò Gnocato*

Abstract

This paper studies the optimal conduct of monetary policy in the presence of heterogeneous exposure to energy price shocks between the employed and the unemployed, as it is documented by data from the euro-area Consumer Expectations Survey: higher energy prices weigh more on the unemployed, who consume less and devote a higher proportion of their consumption to energy. I account for this evidence into a tractable Heterogeneous-Agent New Keynesian (HANK) model with Search and Matching (S&M) frictions in the labour market, and energy as a complementary input in production and as a non-homothetic consumption good: energy price shocks weigh more on the jobless, who consume less due to imperfect unemployment insurance and, since preferences are non-homothetic, devote a higher share of this lower consumption to energy. Households' heterogeneous exposure to rising energy prices induces an endogenous trade-off for monetary policy, whose optimal response involves partly accommodating core inflation so as to indirectly sustain employment and, therefore, prevent workers from becoming more exposed to the shock through unemployment.

JEL Classification: E21, E24, E31, E32, E52.

Keywords: heterogeneous agents, new Keynesian, unemployment risk, energy shocks, optimal monetary policy, endogenous trade-off.

DOI: 10.32057/0.TD.2024.1450

^{*} Bank of Italy, Economic Outlook and Monetary Policy Directorate.

The views expressed in this paper are those of the author and should not be attributed to the Bank of Italy or the Eurosystem. I am extremely grateful to Tommaso Monacelli and Antonella Trigari for their advice. I thank Guido Ascari, Andrea Colciago, Basile Grassi, Philip R. Lane, Stefano Neri, Andrea Papetti, Nicola Pavoni, Massimiliano Pisani, Amalia Repele, Marianna Riggi, Tiziano Ropele, Daniele Siena, Luigi Federico Signorini, Giordano Zevi and Roberta Zizza for insightful comments and discussions. I am furthermore indebted to several seminar and conference participants for their valuable suggestions. All errors are my own.

1 Introduction

Since the end of 2021 and through mid-2023, euro-area economies have had to cope with exceptionally high inflation, both energy and overall, and depressed real wages (Figure 1). The recent surge in energy prices has not only depressed workers' real income but has also had distributional consequences. Indeed, this type of shock affects consumers heterogeneously: as energy is a necessity good, poorer households devote a larger share of their income to energy consumption and are hit harder when the price increases.¹ Given that workers suffer substantial losses in income and consumption during unemployment spells,² the latter constitute a channel through which households can become more exposed to rising energy prices. Exploiting data from the euro-area Consumer Expectations Survey, I document how the unemployed allocate a significantly higher share of their overall expenditure in goods and services to energy-intensive consumption (utilities and transport services) compared to the employed. Moreover, this disparity widened further after the shock hit, despite the fact that most countries implemented several measures aimed at relieving the pressure on poorer and, hence, more exposed households (Arregui et al., 2022).

In light of these facts, it appears crucial to understand how monetary authorities can influence household exposure to energy price shocks via the unemployment rate, given that the latter is one of the critical indicators guiding monetary policy decisions. This paper studies how monetary policy should optimally react to these shocks through the lens of a tractable Heterogeneous-Agent New Keynesian (HANK) model with Search and Matching (S&M) frictions in the labour market and energy as a complementary input in production and a non-homothetic consumption good, wherein employed and unemployed workers are heterogeneously affected by energy price shocks due to the joint presence of imperfect unemployment insurance and non-homothetic preferences. More specifically, and differently from the homothetic case previously considered in the HANK-S&M literature,³ consumption is limited to what remains after covering subsistence energy needs. Hence, lower consumption will be associated with a higher energy consumption share. In other words, due to the presence of imperfect unemployment insurance, the unemployed are forced to consume less, and, in addition, non-homothetic energy consumption implies that they devote a higher share of their consumption to energy compared to the employed. Therefore, higher energy prices end up weighing more on the jobless and, due to the presence of a frictional labour market, employed

¹Recent studies that have noted the distributional impact of energy price surges due to their effect on heterogeneous consumption baskets include Ari et al. (2022); Bachmann et al. (2022); Battistini et al. (2022); Curci et al. (2022).

^{2}Various studies have documented these consumption losses to range from 14% to 26% (Den Haan et al., 2018).

 $^{^{3}}$ See, for instance, Ravn and Sterk (2021) and Challe (2020).



Figure 1: INFLATION AND REAL WAGES IN THE EURO AREA

Notes: The left panel of the figure reports quarterly averages of monthly year-on-year percentage changes in the euro-area Harmonised Index of Consumer Prices (HICP) and in its energy and non-energy components, while the middle panel reports the weighted contributions of the two components to the changes of the overall index; the HICP "energy" classification includes electricity, gas, liquid fuels, solid fuels, heat energy, and fuels and lubricants for personal transport equipment; latest observation: 2023 Q4. The right panel reports the average compensation per employee in the euro area, deflated using the HICP and normalised at 100 in 2019 Q4; latest observation: 2023 Q3. *Sources:* ECB and Eurostat.

workers also can transition endogenously to unemployment and hence become more exposed to the shock. In turn, the monetary authority can indirectly mitigate the share of more exposed households by trading off higher inflation with lower unemployment.

I show analytically, by relying on a simple positive illustrative case, how rising energy prices induce firms to reduce activity and employment when energy is a complementary input in production. Moreover, in the presence of imperfect insurance, this response is amplified through both an unemployment risk channel and the heterogeneous exposure to the shock between the employed and the unemployed. First, as unemployment risk increases and households cannot perfectly insure against it, precautionary saving desires also increase, and aggregate demand is depressed. This, in turn, sets in motion the feedback loop already highlighted, among others, by Ravn and Sterk (2021), whereby production falls even further, feeding back to even greater unemployment risk, and so on. Second, and on top of this, due to the presence of non-homothetic preferences, the increase in real energy prices raises the consumption losses upon unemployment, strengthening unemployment fears and precautionary saving motives, further depressing aggregate demand and employment.

These two sources of amplification of the unemployment response to rising energy prices unemployment risk and the heterogeneous exposure to the shock— pose, in turn, the normative question of how monetary policy should optimally react. First, by relying on the same simplified case for fully analytical characterisations, I show that fluctuations in real energy prices endogenously induce a monetary policy trade-off between stabilising core inflation and maintaining unemployment at its desired, constrained-efficient level. This trade-off originates as an endogenous wedge between natural unemployment (the level consistent with stable core inflation) and constrained-efficient unemployment (the welfare-maximising level). Intuitively, the presence of subsistence energy consumption and imperfect unemployment insurance jointly imply that employed and unemployed workers are heterogeneously exposed to energy price shocks. Consequently, rising energy prices induce an increase in the consumption losses upon unemployment. As a result, it is optimal for the monetary authority to partly accommodate core inflation in order to indirectly sustain employment. This helps to limit the increase in unemployment induced by the shock hitting producers, thereby preventing households from becoming more exposed to the shock through unemployment.

Second, I numerically explore the mechanism for the general case, confirming the analytical results: compared to a policy rule aimed at fully stabilising core inflation, the optimal policy is able to achieve a smaller increase in unemployment at the cost of partly accommodating core inflation; this rise in inflation is, however, smaller than if unemployment were instead to be fully stabilised. It is also confirmed that the trade-off arises due to the presence of heterogeneous exposure to the energy price shock between the employed and the unemployed: considering the counterfactual case of homothetic preferences and hence homogeneous direct exposure to the shock, the optimal policy coincides with one aimed at fully stabilising core inflation. In both cases, while the monetary authority reacts more or less aggressively to core inflation, the energy component of inflation is instead fully looked through under the optimal policy. Intuitively, only core inflation endogenously matters for the hiring incentives of producers, being only the core goods produced within the economy. In contrast, energy is treated as a non-produced good, with its (real) price evolving exogenously. This simplifying assumption, in line with the approach of Blanchard and Galí (2007), concisely captures the situation faced by an energy-importing economy such as the euro area.

Related Literature. This paper contributes, first, to the macroeconomic literature on the impact of energy shocks on producers and consumers: earlier studies (Blanchard and Galí, 2007; Bodenstein et al., 2008; Montoro, 2012; Blanchard and Riggi, 2013) have considered representative-agent settings, providing the normative insight that central banks should react aggressively to core inflation but fully accommodate energy price changes (Natal, 2012). My paper moves forward in this context by making a case for not acting too hard on core inflation, as excess unemployment would make heterogeneously exposed households worse off. Gagliardone and Gertler (2023), still using a representative-agent setting with oil as a complementary good in consumption and complementary input in production, show that the recent inflation surge was mainly accounted for by a combination of oil price shocks

and accommodative monetary policy. In their framework, which also allows for unemployment, a rationale for part of such accommodation is the fact that monetary policy trade-offs arise due to real wage rigidity, while in my heterogeneous-household setting they arise not only due to this but, first and foremost, because energy shocks weigh more on the unemployed. Recent studies have also taken household heterogeneity into account, for instance in Two-Agent New Keynesian (TANK) frameworks à la Debortoli and Galí (2017): Chan et al. (2023) investigate how energy shocks impacting producers can feed back to depressed demand, but without considering how energy shocks *directly* affect consumers, an aspect which I explicitly model with liquidity-constrained households being more exposed to these shocks due to non-homothetic energy consumption. Corsello and Riggi (2023) develop a TANK framework with energy in production and heterogeneously exposed households, as the hand-to-mouth devote a higher (constant) share of their consumption expenditure to energy,⁴ hence ending up experiencing a higher inflation rate. Their analysis is positive rather than normative, offering a historical decomposition of inflation inequality into its main drivers, including the monetary policy stance. Kharroubi and Smets (2023) introduce a minimum subsistence requirement for energy consumption to explore optimal *fiscal* responses to an energy shock in a two-agent framework, and advocate subsidising poorer, constrained households while taxing those that are rich and unconstrained. My analysis delves into optimal monetary policy, keeping fiscal policy constant. Moreover, while these Two-Agent frameworks assume a constant fraction of hand-to-mouth households, in my setting, the fraction of agents ending up being liquidity-constrained is endogenous and microfounded, as it is captured by the unemployment rate. In such context, Kharroubi and Smets' fiscal approach cannot be easily translated since it would involve taxing the employed and subsidising the unemployed, potentially dampening job-search incentives. Other recent studies have accounted for household heterogeneity within quantitative HANK models à la Kaplan et al. (2018): Pieroni (2023) introduces non-homothetic preferences in a similar fashion as I do, but in a rich quantitative model with energy consumption by both households and firms, thus relying only on numerical solutions and focusing on positive analysis. In my work, I consider instead a specific dimension of household heterogeneity —stemming from imperfect unemployment insurance— which allows me to characterise and investigate analytically the mechanisms at the source of monetary policy trade-offs with their normative implications.

Second, I build on the recent literature studying the aggregate effects of incomplete markets in

⁴Their setting aligns with Känzig's (2023), who rationalises his findings that a tighter carbon pricing regime results in higher energy prices, disproportionately affecting the poor due to their higher energy share. Given that Känzig's shocks are, essentially, energy price shocks, their effects can also be rationalised within my framework.

the form of imperfectly insurable unemployment risk within tractable HANK-S&M models (Challe et al., 2017; Ravn and Sterk, 2017, 2021; Den Haan et al., 2018; McKay and Reis, 2021). My work relates in particular to Challe (2020), who has investigated the optimal conduct of monetary policy within this class of models in response to productivity shocks and generic cost-push shocks.

I contribute to these strands of literature, in the first instance, by modelling energy as an input in both production and consumption within a tractable HANK-S&M framework and innovating on the existing studies by uncovering a novel precautionary saving motive and source of monetary policy trade-off, both arising endogenously due to non-homothetic preferences and imperfect insurance, focusing on normative implications analytically as well as quantitatively. In this vein, my work also relates to Acharya et al. (2023), who study how (optimal) monetary policy can mitigate consumption risk arising from households' unequal exposure to aggregate shocks in a tractable HANK model. In a similar spirit, Smirnov (2022) shows numerically, using a rich quantitative HANK model, how optimal policy aims to mitigate this unequal exposure.

Roadmap. The remainder of this paper is structured as follows. Section 2 provides motivating evidence. Section 3 describes the model, and Section 4 characterises its constrained-efficient allocation. Section 5 focuses on the linearised version of the model and deals with positive analysis, while normative analysis is performed in Section 6. Section 7 concludes.

2 Motivating Evidence

In this section, I exploit the microdata from the European Central Bank's Consumer Expectations Survey (CES) to document some facts about the heterogeneity between the employed and the unemployed in their exposure to energy price shocks. The CES covers the six largest economies in the euro area (Germany, France, Italy, Spain, the Netherlands, and Belgium) over the period since April 2020. On a quarterly basis, it collects information on individual monthly spending in twelve major categories, including energy-intensive ones (utilities and transport services), as well as their employment situation.⁵ This allows to compute an individual-level measure of exposure to energy price shocks as the share of energy-intensive categories in overall expenditure on goods and services.

Figure 2 reports the average energy shares of the employed and the unemployed over time,

⁵The "utilities" category includes water, sewerage, electricity, gas, heating oil, phone, cable, internet; the "transport" category includes fuel, car maintenance, public transportation fares. Information about the employment situation is available from the October 2020 wave of the survey. Appendix A provides a more detailed description of the data.



Figure 2: Energy Shares of the Employed and the Unemployed

Notes: The figure reports the average shares of energy-intensive consumption expenditure over time for employed and unemployed individuals interviewed in the CES, obtained in regressions of the form $e_{it} = \beta_0 + \beta_1 une_{it} + X_{it} \gamma + \varepsilon_{it}$, where e_{it} is the energy share of individual *i* at time *t*, $une_{it} = 1$ if the individual is employed and zero if employed, and X_{it} controls for individual-specific characteristics (age, gender, birth abroad, education, presence of a partner). The vertical bars indicate 95% confidence intervals.

2021 Q3 - 2022 Q2

2020 Q3 - 2021 Q2

controlling for individual-specific observable characteristics.⁶ Between late 2020 and early 2021 —before the surge in energy prices— the share was around 18% for the employed and about 3 percentage points higher for the unemployed. As energy prices surged, the share rose to nearly 20% for the employed and around 4 percentage points higher for the unemployed. Therefore, even if many countries have implemented fiscal measures targeted at mitigating the impact of the shock on vulnerable households (Arregui et al., 2022), the energy share for the unemployed has increased by around 1 percentage point more than for the employed.

To investigate in more depth the presence of different consumption patterns between the employed and the unemployed, in particular on energy-intensive consumption, I then focus on the period before the energy price surge (2020 Q3 to 2021 Q2) and look at average differences in the expenditure shares of the unemployed relative to that of the employed as well as at average percentage differences in overall consumption, utilities, and transport services expenditure.

As shown in Table 1, controlling for individual-specific observable characteristics (age, gender, birth abroad, education, presence of a partner), the employed devoted, on average, 11.2% of their

⁶These include age, gender, birth abroad, education, presence of a partner. See Appendix A for more details.

Table 1: Energy-intensive Consumption and Unemploymen

		$\ln(consum)$	nption)	$\ln(ut)$	ilities)	$\ln(trans)$	ports)	$\ln(energy)$	
unemployed -0.247		17	-0.043		-0.359		-0.124		
		[-0.283,-0.212] [-0.		[-0.082	2,-0.005]	5] [-0.465,-0.252]		[-0.166,-0.08	2]
(b) Expenditure Shares									
			utilities	s share	transpo	rts share	total	energy share	
	unemployed 0.03		31	-0.002		0.029			
			[0.025,	0.037]	[-0.005	5,0.001]	[0.0	[022, 0.036]	
employed (baseline)		baseline)	0.112		0.	070		0.182	
			[0.111,	0.113]	[0.069]	,0.071]	[0.1]	180, 0.183]	

(a) Expenditure

Notes: The table reports results of regressions of the form $y_{it} = \beta_0 + \beta_1 une_{it} + X_{it} \gamma + \varepsilon_{it}$, run on 20899 observations from the CES for the period from 2020 Q3 to 2021 Q2. y_{it} is, alternatively, the share of energy-intensive items (utilities and transport services) in the overall consumption expenditure of individual *i* at time *t*, the log of their overall consumption expenditure, log expenditure on utilities, transport services, and their sum. $une_{it} = 1$ when individual *i* is unemployed at time *t* and 0 when employed. X_{it} controls for individual-specific characteristics (age, gender, birth abroad, education, presence of a partner). 95% confidence intervals are reported in brackets.

overall consumption expenditure to utilities, while this share was 3.1 percentage points higher for the unemployed. This is due to the fact that while overall consumption expenditure was $\exp\{-0.247\} - 1 \simeq 22\%$ lower for the unemployed compared to the employed, expenditure on utilities was only around 4% lower. Moreover, despite the fact that expenditure on transport services was around 30% lower for the unemployed, their expenditure share of energy-intensive consumption (utilities plus transport services) was nevertheless around three percentage points higher than that of the employed, for whom it was slightly above 18%.

3 The Model

The model tracks a sticky-price economy with uninsurable unemployment risk, along the lines of Challe (2020), enriched by introducing energy as an input both in production and in the consumption bundle. Following Blanchard and Galí (2007) and most of the subsequent literature on the macroeconomic effects of oil and energy shocks, energy is regarded as a non-produced input whose real price evolves according to an exogenous stochastic process. On the household side, there is a unit measure of workers who, due to their inability to borrow, cannot perfectly insure against the risk of becoming unemployed, and who derive utility from a basket of a core consumption good and energy. The latter is consumed above a subsistence level, implying that poorer households (i.e. the unemployed, in this setting) will devote a higher share of their overall consumption expenditure to energy. Additionally, there is a measure ν of risk-neutral firm owners who simply collect and consume hand-to-mouth the dividends arising in the production sector, net of fiscal transfers.

On the producer side, the core good is assembled by aggregating differentiated varieties from a monopolistically competitive wholesale sector. Each wholesaler uses, in turn, energy and labour services as inputs. These services are supplied competitively by intermediaries who hire labour from households in a market with standard S&M frictions à la Mortensen and Pissarides (1994).⁷

3.1 Households

Workers face idiosyncratic income risk from being either employed or unemployed: due to imperfect insurance, the employed and the unemployed will have different consumption levels at equilibrium.

The consumption basket of each worker type $i \in \{n, u\}$ (employed, unemployed), c_t^i , is defined as a Stone-Geary aggregator of a core consumption good, g_t^i , and energy, e_t^i , which is consumed above a subsistence level $\xi > 0$,

$$c_t^i = (g_t^i)^{(1-\omega_e)} (e_t^i - \xi)^{\omega_e}$$
(1)

where $\omega_e \in (0, 1)$ is the quasi-share of energy. Allowing for the presence of the subsistence level ξ will imply, consistently with the empirical evidence, that energy consumption demand is inelastic and that poorer households (i.e. the unemployed, in this setting) will devote a higher share of their income to energy consumption.⁸

The expected lifetime utility of a currently employed (resp. unemployed) worker can be formulated recursively as follows

$$U_t^n = \ln (c_t^n) + \beta \mathbb{E}_t \left[(1 - \lambda_{t+1}) U_{t+1}^n + \lambda_{t+1} U_{t+1}^u \right]$$
$$U_t^u = \ln (c_t^u) + \beta \mathbb{E}_t \left[f_{t+1} U_{t+1}^n + (1 - f_{t+1}) U_{t+1}^u \right]$$

⁷Introducing labour market frictions in a separate, upstream sector is isomorphic to considering these frictions affecting wholesalers who directly hire labour from households.

⁸See Appendix B.2 for more details about the properties of consumer demand in the presence of the non-homotheticities of Stone-Geary preferences.

where λ is the transition rate from employment to unemployment, while f is the job-finding rate. Each worker type maximises expected lifetime utility subject to (1) and

$$\begin{cases} P_{g,t} g_t^i + P_{e,t} e_t^i + B_t^i = Y_t^i + (1 + i_{t-1}) B_{t-1}^i \\ B_t^i \ge 0 \end{cases}$$

where $P_{e,t}$ is the price of energy and $P_{g,t}$ is the price of the core consumption good; B_t denotes the nominal amount of 1-period bonds held at the end of period t, and i_t is the nominal return on these assets. Y_t^i is equal either to W_t (the nominal wage income a worker receives when employed) or Δ_t (nominal home production when unemployed).

Demands for the core consumption good and energy are given by

$$g_t^i = (1 - \omega_e) \left(\frac{P_t}{P_{g,t}}\right) c_t^i \tag{2}$$

$$e_t^i = \omega_e \left(\frac{P_t}{P_{e,t}}\right) c_t^i + \xi \tag{3}$$

where

$$P_t = \left(\frac{P_{g,t}}{1 - \omega_e}\right)^{(1 - \omega_e)} \left(\frac{P_{e,t}}{\omega_e}\right)^{\omega_e} \tag{4}$$

represents the overall consumer price index (CPI). Letting $\pi_{t+1} := \frac{P_{t+1}}{P_t} - 1$, to a first order

$$\pi_t \simeq (1 - \omega_e) \,\pi_{g,t} + \omega_e \,\pi_{e,t} \,. \tag{5}$$

Euler conditions for employed and unemployed workers are, respectively,

$$\frac{1}{c_t^n} \ge \beta \mathbb{E}_t \left\{ \left(\frac{1+i_t}{1+\pi_{t+1}} \right) \left[(1-\lambda_{t+1}) \frac{1}{c_{t+1}^n} + \lambda_{t+1} \frac{1}{c_{t+1}^u} \right] \right\}$$
(6)

$$\frac{1}{c_t^u} \ge \beta \mathbb{E}_t \left\{ \left(\frac{1+i_t}{1+\pi_{t+1}} \right) \left[f_{t+1} \frac{1}{c_{t+1}^n} + (1-f_{t+1}) \frac{1}{c_{t+1}^u} \right] \right\}$$
(7)

each holding with strict inequality if the agent is liquidity-constrained (i.e. wishing to borrow) and with equality otherwise. Due to the presence of non-homothetic preferences, captured by $\xi > 0$, consumption does not equal real overall expenditure on goods and is instead given by

$$c_t^i = \frac{X_t^i}{P_t} - \frac{P_{e,t}}{P_t} \xi$$

where $X_t^i = P_{g,t} g_t^i + P_{e,t} e_t^i$. Intuitively, households derive utility only from what is left over after spending on energy for subsistence purposes. This form of non-homotheticity has the advantage of linking the presence of different consumption patterns directly to the inability to perfectly insure against job losses, which will force the unemployed to spend less on consumption at equilibrium.

Employment Dynamics. Worker transitions between employment and unemployment give rise to the following laws of motion for the stocks of employed (n_t) and unemployed (u_t) workers

$$n_{t+1} = (1 - \lambda_{t+1}) n_t + f_{t+1} u_t$$
$$u_{t+1} = (1 - f_{t+1}) u_t + \lambda_{t+1} n_t$$

where $u_t = 1 - n_t$. Being $s_{t+1} = u_t + \rho n_t$ the stock of effective searchers, $\lambda_{t+1} = \rho (1 - f_{t+1})$, where ρ is the separation rate. In other words, a worker employed in the current period faces a risk of being unemployed in the next period given by λ_{t+1} , the joint probability of undergoing separation, and not being able to find another job subsequently.

3.2 Producers

The core consumption good is produced under perfect competition by aggregating a continuum of wholesale goods with a constant elasticity of substitution technology

$$Y_t = \left(\int_0^1 y_t(k)^{\frac{\varepsilon-1}{\varepsilon}} dk\right)^{\frac{\varepsilon}{\varepsilon-1}} .$$
(8)

Therefore, demand for variety k is given by

$$y_t(k) = Y_t \left[\frac{P_t(k)}{P_{g,t}}\right]^{-\varepsilon}$$
(9)

where $P_{g,t} = \left(\int_0^1 P_t(k)^{1-\varepsilon} dk\right)^{1/(1-\varepsilon)}$, and $\varepsilon > 1$.

Wholesalers

Each wholesaler is a monopolistic supplier of the variety k it produces using labour services as well as energy, combined in fixed proportions according to the following production function

$$y_t(k) = \min\left\{\frac{l_t(k)}{1-\gamma_e}, \frac{e_t(k)}{\gamma_e}\right\}$$

where $y_t(k)$ is the amount of variety k produced, $e_t(k)$ and $l_t(k)$ are respectively the amounts of energy and intermediate inputs used in production by wholesaler k, and $\gamma_e \in (0, 1)$ captures the factor proportion of energy relative to labour services.

Demand for inputs from wholesalers will then be given by

$$l_t(k) = (1 - \gamma_e) y_t(k)$$
(10)

$$e_t(k) = \gamma_e \, y_t(k) \tag{11}$$

and the nominal marginal cost by

$$MC_t = (1 - \gamma_e) \Phi_t + \gamma_e P_{e,t} \tag{12}$$

where Φ_t denotes the price of labour services and $P_{e,t}$ is the energy price.

Moreover, wholesale firms are assumed to face Calvo pricing frictions, with θ being the probability that a wholesale firm cannot reset its price. Therefore, the optimal reset price, P_t^* , satisfies

$$p_{g,t}^* = \frac{P_t^*}{P_{g,t}} = \frac{\mathcal{Y}_t}{\mathcal{Z}_t} \tag{13}$$

where \mathcal{Y}_t and \mathcal{Z}_t obey the following recursions

$$\mathcal{Y}_{t} = (1 - \tau_{y}) \left(\frac{\varepsilon}{\varepsilon - 1}\right) \frac{MC_{t}}{P_{t}} Y_{t} + \theta \beta \mathbb{E}_{t} \left[(1 + \pi_{g, t+1})^{\varepsilon} \mathcal{Y}_{t+1}\right]$$
$$\mathcal{Z}_{t} = \frac{P_{g, t}}{P_{t}} Y_{t} + \theta \beta \mathbb{E}_{t} \left[(1 + \pi_{g, t+1})^{(\varepsilon - 1)} \mathcal{Z}_{t+1}\right]$$

and τ_y is a production subsidy, financed through a lump-sum tax on firm owners, aimed at offsetting the steady-state distortion due to monopolistic competition.

Given Calvo pricing frictions and symmetry in the wholesale sector, gross inflation in the core consumption good price, $1 + \pi_{g,t} = P_{g,t}/P_{g,t-1}$, evolves according to

$$1 + \pi_{g,t} = \left[\frac{1}{\theta} - \left(\frac{1-\theta}{\theta}\right) \left(p_{g,t}^*\right)^{(1-\varepsilon)}\right]^{1/(\varepsilon-1)} .$$
(14)

Real dividends collected by firm owners from wholesaler k are then given by

$$d_t^W(k) = \frac{1}{P_t} \left[P_t(k) \, y_t(k) - (1 - \tau_y) \, \Phi_t \, l_t(k) - (1 - \tau_y) \, P_{e,t} \, e_t(k) \right] \tag{15}$$

and aggregate real dividends from the wholesale sector by

$$d_t^W = \int_0^1 d_t^W(k) \, dk = Y_t \, \{ p_{g,t} - (1 - \tau_y) \, [(1 - \gamma_e) \, \varphi_t + \gamma_e \, p_{e,t}] \, \mathcal{D}_t \}$$
(16)

where $p_{g,t} = P_{g,t}/P_t$, $\varphi_t = \Phi_t/P_t$, and

$$\mathcal{D}_t := \int_0^1 \left[\frac{P_t(k)}{P_{g,t}} \right]^{-\varepsilon} dk = (1-\theta) \left(p_{g,t}^* \right)^{-\varepsilon} + \theta \prod_{g,t}^{\varepsilon} \mathcal{D}_{t-1}$$
(17)

is an index of price dispersion among wholesalers.

Labour Intermediaries

Intermediaries hire labour from households in a frictional market. These frictions are summarised by an aggregate matching function, which is assumed to take a Cobb-Douglas form with constant returns to scale,

$$m_t = s_t^{\alpha} \, v_t^{(1-\alpha)} \tag{18}$$

where m_t denotes the total amount of formed matches, v_t is the total amount of vacancies, $\alpha \in (0, 1)$, and $s_t = u_{t-1} + \rho n_{t-1}$ is the total amount of searching workers, given by those workers who were unemployed plus those workers who were employed but experience separation (at rate ρ). The job-finding and vacancy-filling rates are given, respectively, by

$$f_t = \frac{m_t}{s_t} \,, \tag{19}$$

$$q_t = \frac{m_t}{v_t} = f_t^{\frac{\alpha}{\alpha-1}} \,. \tag{20}$$

Active matches produce one unit of output at each period. The value of a match is

$$J_{t} = (1 - \tau_{z}) (\varphi_{t} - w_{t} + S) + \beta (1 - \rho) \mathbb{E}_{t} (J_{t+1})$$
(21)

where $w_t = W_t/P_t$ is the real wage rate, and $\varphi_t = \Phi_t/P_t$ is the price of labour services in terms of the final good. τ_z and S are fiscal instruments that will be aimed at offsetting the steady-state distortions arising from labour market frictions and imperfect unemployment insurance.

When matches break up, inactive intermediaries post vacancies at a cost of κ units of the final good per vacancy per period, and each vacancy is filled with probability q_t . Free entry into vacancy posting gives rise to the following job creation condition

$$f_t^{\frac{\alpha}{1-\alpha}} = \frac{1-\tau_z}{\kappa} \left(\varphi_t - w_t + S\right) + \beta \left(1-\rho\right) \mathbb{E}_t \left(f_{t+1}^{\frac{\alpha}{1-\alpha}}\right) \,. \tag{22}$$

Aggregate real dividends collected by firm owners from labour intermediaries are then given by

$$d_t^I = n_t (1 - \tau_z) (\varphi_t - w_t + S) - \kappa v_t.$$
(23)

3.3 Equilibrium

3.3.1 Market Clearing

Energy Market. Following Blanchard and Galí (2007), energy is modelled as a non-produced input whose real price is assumed to evolve exogenously according to

$$\ln(p_{e,t}) = \rho_e \,\ln(p_{e,t-1}) + \epsilon_t^e \tag{24}$$

where $\rho_e \in [0, 1)$ and $\epsilon_t^e \sim iid(0, \sigma_e^2)$. In other words, the aggregate market demand for energy from households and firms is assumed to be always cleared at the exogenous real price $p_{e,t}$.

Labour Market. The total supply of labour services is given by the measure of active matches, n_t , while total demand can be obtained by aggregating over wholesalers. Given (9) and (10), we have

$$\int_0^1 l_t(k) = \int_0^1 (1 - \gamma_e) \, y_t(k) = (1 - \gamma_e) \, Y_t \, \mathcal{D}_t \,.$$
(25)

Therefore, labour market clearing requires

$$n_t = (1 - \gamma_e) Y_t \mathcal{D}_t.$$
⁽²⁶⁾

Good Market. The supply of the core consumption good is given by Y_t . Its total demand, $n_t g_t^n + (1 - n_t) g_t^u + \nu g_t^o$, is obtained by aggregating over employed and unemployed workers and firm owners. Given (2), market clearing then requires

$$(1 - \omega_e) \left[n_t c_t^n + (1 - n_t) c_t^u + \nu c_t^o \right] = p_{g,t} Y_t$$
(27)

where c_t^n , c_t^u , c_t^o are, respectively, the final consumption baskets of employed and unemployed workers and firm owners.

3.3.2 Household Consumption and Zero Liquidity Property

Workers. Given the zero-debt limit households face, the supply of assets is always zero at equilibrium, no asset trade actually takes place, and all households turn out to spend all their current income on consumption. Therefore, letting $\delta_t = \Delta_t/P_t$ denote real home production when unemployed, the consumption bundles of the employed and the unemployed are, respectively,

$$c_t^n = w_t - p_{e,t}\,\xi\tag{28}$$

$$c_t^u = \delta_t - p_{e,t}\,\xi\tag{29}$$

from which it is immediate to see the following,

Proposition 1. If $\delta_t < w_t$, then $c_t^u < c_t^n$ and a rise in the real price of energy implies that the consumption of the unemployed falls proportionately more than the consumption of the employed:

$$\frac{\partial c_t^u / c_t^u}{\partial p_{e,t} / p_{e,t}} < \frac{\partial c_t^n / c_t^n}{\partial p_{e,t} / p_{e,t}} < 0$$

In light of (28) and (29), the Euler conditions of the employed and the unemployed are, respectively,

$$1 \ge \beta \mathbb{E}_t \left\{ \left(\frac{1+i_t}{1+\pi_{t+1}} \right) \left[(1-\lambda_{t+1}) \left(\frac{w_t - p_{e,t} \xi}{w_{t+1} - p_{e,t+1} \xi} \right) + \lambda_{t+1} \left(\frac{w_t - p_{e,t} \xi}{\delta_{t+1} - p_{e,t+1} \xi} \right) \right] \right\}$$
(30)

$$1 \ge \beta \mathbb{E}_t \left\{ \left(\frac{1+i_t}{1+\pi_{t+1}} \right) \left[f_{t+1} \left(\frac{\delta_t - p_{e,t} \xi}{w_{t+1} - p_{e,t+1} \xi} \right) + (1 - f_{t+1}) \left(\frac{\delta_t - p_{e,t} \xi}{\delta_{t+1} - p_{e,t+1} \xi} \right) \right] \right\}$$
(31)

each holding with strict inequality if the household is liquidity-constrained (i.e. wishing to borrow) and with equality otherwise.

It can be shown that at equilibrium, in the steady state neighbourhood, the Euler condition of employed workers holds with equality while that of the unemployed holds with inequality.⁹ Formally:

Proposition 2. When $\delta < w$, the steady-state Euler conditions of employed and unemployed ⁹See Ravn and Sterk (2021) and Challe (2020) for more details.

workers are, respectively,

$$1 = \beta \left(\frac{1+i}{1+\pi}\right) \left[(1-\lambda) + \lambda \left(\frac{w-p_e \xi}{\delta - p_e \xi}\right) \right]$$
$$1 > \beta \left(\frac{1+i}{1+\pi}\right) \left[f \left(\frac{\delta - p_e \xi}{w - p_e \xi}\right) + (1-f) \right]$$

Proof. See Appendix B.3.

In other words, at equilibrium, employed workers wish to precautionarily save, but as the unemployed are impeded from borrowing, no one is able to issue the assets that would allow this precautionary saving desire to be actually satisfied. Hence, no asset trade eventually takes place at equilibrium. This zero-liquidity property allows to price precautionary saving *desires* without the need to track a full wealth distribution over time.

Firm Owners. The final consumption basket of firm owners equals their after-tax real income, net of subsistence energy needs, $c_t^o = (d_t^I + d_t^W + \tau_t) - \xi p_{e,t}$, where

$$\tau_t = -\tau_y \left[(1 - \gamma_e) \varphi_t + \gamma_e p_{e,t} \right] \mathcal{D}_t Y_t + \tau_z n_t \left(\varphi_t - w_t \right) - n_t \left(1 - \tau_z \right) S.$$
(32)

Therefore, at equilibrium, firm owner consumption is

$$c_t^o = \left(\frac{p_{g,t} \mathcal{D}_t^{-1} - \gamma_e p_{e,t}}{1 - \gamma_e} - w_t\right) n_t - \kappa v_t - \xi p_{e,t}$$
(33)

where, given the equilibrium law of motion of employment, $n_t = (1-\rho) n_{t-1} + [1 - (1-\rho) n_{t-1}]^{\alpha} v_t^{1-\alpha}$, v_t can be expressed as

$$v_t = \left\{ \frac{n_t - (1 - \rho) n_{t-1}}{[1 - (1 - \rho) n_{t-1}]^{\alpha}} \right\}^{\frac{1}{1 - \alpha}}.$$
(34)

4 Constrained-Efficient Allocation

The constrained-efficient allocation is the solution to the problem of maximising the aggregate welfare of the economy, taking into account the decentralised equilibrium reactions by consumers and producers. Considering a utilitarian planner who attaches equal weights to the utility of each household, the flow welfare of the economy is given by

$$U_t = n_t \ln(c_t^n) + (1 - n_t) \ln(c_t^u) + \nu c_t^o$$
(35)

where c_t^n , c_t^u , and c_t^o are given, respectively, by (28), (29), and (33). Therefore, the constrainedefficient allocation is the solution to

$$W(\mathcal{D}_{t-1}, n_{t-1}, p_{e,t}) = \max_{\{p_{g,t}^*, w_t, n_t\}} \{ U_t + \beta \mathbb{E}_t[W(\mathcal{D}_t, n_t, p_{e,t+1})] \}$$
(36)

subject to (14) and (17).

Now, $p_{g,t}^* = 1$ at every period is optimal, as it implies $\pi_{g,t} = 0$, thereby ensuring that $\mathcal{D}_t = 1$: zero inflation eliminates the dispersion in wholesale prices, that would imply lower output from given inputs, as can be seen from (26). Therefore, the problem reduces to

$$W(n_{t-1}, p_{e,t}) = \max_{\{w_t, n_t\}} \{ U_t + \beta \mathbb{E}_t[W(n_t, p_{e,t+1})] \} .$$
(37)

The first-order condition with respect to w_t gives the following constrained-efficient wage rate

$$w_t^* = \frac{1}{\nu} + \xi \, p_{e,t} \,. \tag{38}$$

In other words, the constrained-efficient allocation involves risk-neutral firm owners insuring workers against fluctuations in their equilibrium consumption bundle as given in (28). Due to the need to consume energy for subsistence purposes, a constant wage rate is not sufficient for the purpose in this setting, in contrast with Challe (2020). Insuring workers against consumption fluctuations involves, instead, the constrained-efficient wage rate adjusting upward (downward) in response to positive (negative) shocks to real energy prices.

The first-order condition with respect to n_t gives the following forward recursion for the constrained-efficient job-finding rate

$$f_{t}^{*\frac{\alpha}{1-\alpha}} = \frac{1-\alpha}{\kappa} \left[\frac{p_{g,t} - \gamma_{e} \, p_{e,t}}{1-\gamma_{e}} - w_{t}^{*} + \frac{1}{\nu} \ln\left(\frac{w_{t}^{*} - \xi \, p_{e,t}}{\delta_{t} - \xi \, p_{e,t}}\right) \right] + \beta \left(1-\rho\right) \mathbb{E}_{t} \left[f_{t+1}^{*\frac{\alpha}{1-\alpha}} \left(1-\alpha \, f_{t+1}^{*}\right) \right]$$
(39)

from which the constrained-efficient level of employment can be derived by exploiting the law of motion $n_t^* = (1 - \rho) n_{t-1}^* + f_t^* [1 - (1 - \rho) n_{t-1}^*].$

4.1 Steady State

First of all, the constrained-efficient steady state must entail zero inflation, requiring in turn $p_g^* = 1$, i.e. $\varphi = \frac{1}{1-\tau_y} \left(\frac{\varepsilon-1}{\varepsilon}\right) \left(\frac{1}{1-\gamma_e}\right) p_g - \left(\frac{\gamma_e}{1-\gamma_e}\right) p_e$. Comparing the steady state counterparts of (22) and (39), which read respectively

$$f^{\frac{\alpha}{1-\alpha}} = \frac{1-\tau_z}{\kappa \left[1-\beta \left(1-\rho\right)\right]} \left(\varphi - w + S\right)$$
$$f^{*\frac{\alpha}{1-\alpha}} = \frac{1-\alpha}{\kappa \left[1-\beta \left(1-\rho\right) \left(1-\alpha f^{*}\right)\right]} \left[\frac{p_g - \gamma_e p_e}{1-\gamma_e} - w^{*} + \frac{1}{\nu} \ln\left(\frac{w^{*} - \xi p_e}{\delta - \xi p_e}\right)\right]$$

we can then see that the zero-inflation steady state is constrained-efficient provided that $w = w^* = \frac{1}{\nu} + \xi p_e$, and that the tax instruments are such that

$$\tau_y = \frac{1}{\varepsilon}, \quad S = \frac{1}{\nu} \ln\left(\frac{w^* - \xi \, p_e}{\delta - \xi \, p_e}\right), \quad \tau_z = 1 - \frac{(1 - \alpha) \left[1 - \beta \, (1 - \rho)\right]}{1 - \beta \, (1 - \rho) \, (1 - \alpha \, f^*)} \tag{40}$$

where

$$f^* = \left[\frac{1 - \tau_z}{\kappa \left[1 - \beta \left(1 - \rho\right)\right]} \left(\frac{p_g - \gamma_e \, p_e}{1 - \gamma_e} - w^* + S\right)\right]^{\frac{1 - \alpha}{\alpha}}.$$
(41)

Intuitively, the wholesale subsidy, τ_y , optimally offsets the steady-state markup due to monopolistic competition. The tax on labour intermediaries, τ_z , counteracts congestion externalities in their vacancy posting behaviour. The employment subsidy, S, is aimed at avoiding excess unemployment, that hurts households due to the lack of complete insurance.

Given that the tax instruments are assumed to be constant, the constrained-efficient allocation cannot be decentralised outside the zero-inflation steady state.¹⁰ Even if this mix of fiscal instruments were allowed to be time-varying, it is evident from equations (38)–(39) that it would decentralise the constrained-efficient allocation also outside the steady state only when coupled with a rising real wage in response to surging energy prices, an implausible occurrence in reality.

5 The Linearised Model

In order to investigate how the model responds to fluctuations in energy prices, one needs to specify how the home production of the unemployed, δ_t , behaves in comparison to the income of the employed. Following Challe (2020) as a benchmark, I assume that δ_t fluctuates so that the replacement rate δ_t/w_t is constant. Letting the percentage income loss upon unemployment be $\zeta := 1 - \frac{\delta_t}{w_t} = 1 - \frac{\delta}{w}$, we have

$$\delta_t = (1 - \zeta) w_t \,. \tag{42}$$

¹⁰The fact that these tax instruments cannot be adjusted in response to aggregate shocks reflects the idea that fiscal policy is slow to adjust (Acharya et al., 2023).

Now, letting hatted variables denote level deviations from the zero-inflation steady state, and tilde-variables log (or proportional) deviations, the first-order approximate dynamics of the economy are described by the following set of equations

$$\begin{split} \widehat{n}_t &= \frac{\rho}{f + \rho \left(1 - f\right)} \, \widehat{f}_t + (1 - \rho) \left(1 - f\right) \widehat{n}_{t-1} \\ \widehat{u}_t &= -\widehat{n}_t \\ \widehat{y}_t &= \widehat{n}_t \\ \\ \frac{\kappa}{q f} \left(\frac{\alpha}{1 - \alpha}\right) \, \widehat{f}_t &= (1 - \tau_z) \left(\widehat{\varphi}_t - \widehat{w}_t\right) + \beta \left(1 - \rho\right) \frac{\kappa}{q f} \left(\frac{\alpha}{1 - \alpha}\right) \mathbb{E}_t(\widehat{f}_{t+1}) \\ \\ \widehat{mc}_t &= (1 - \gamma_e) \, \widehat{\varphi}_t + \gamma_e \, p_e \, \widetilde{p}_{e,t} \\ \\ \widehat{mc}_{g,t} &= \frac{1}{p_g} \, \widehat{mc}_t + \frac{\omega_e}{1 - \omega_e} \, \widetilde{p}_{e,t} \\ \\ \pi_{g,t} &= \beta \, \mathbb{E}_t(\pi_{g,t+1}) + \Theta \, \widehat{mc}_{g,t} \\ \\ \pi_t &= \pi_{g,t} + \frac{\omega_e}{1 - \omega_e} \left(\widetilde{p}_{e,t} - \widetilde{p}_{e,t-1}\right) \\ \\ \widetilde{I}_t - \mathbb{E}_t(\pi_{t+1}) &= -\frac{\Lambda}{\lambda} \mathbb{E}_t(\widehat{\lambda}_{t+1}) + \frac{(1 + \Xi_w \, \Lambda \, \Psi) \mathbb{E}_t(\widetilde{w}_{t+1}) - \widetilde{w}_t}{1 - \Xi_w} - \Xi_w \, \frac{(1 + \Lambda \, \Psi) \mathbb{E}_t(\widetilde{p}_{e,t+1}) - \widetilde{p}_{e,t}}{1 - \Xi_w} \\ \\ \widehat{\lambda}_t &= -\rho \, \widehat{f}_t \\ \\ \widetilde{p}_{e,t} &= \rho_e \, \widetilde{p}_{e,t-1} + \epsilon_t^e \end{split}$$

where $\Theta = \frac{1-\theta}{\theta} \left(1-\theta \beta\right)$, $\Lambda = \lambda \left[\lambda + \left(\frac{1-\zeta-\Xi_w}{\zeta}\right)\right]^{-1} \in (0,1)$, $\Psi = \frac{1-\Xi_w}{1-\zeta-\Xi_w} > 1$, and $\Xi_w = \frac{p_e \xi}{w} \in [0,1)$ is the steady-state share of subsistence energy expenditure in the income of the employed.

I let the decentralised wage rate evolve as follows¹¹

$$\widetilde{w}_t = -\chi \, \widetilde{p}_{e,t} \tag{43}$$

where the elasticity with respect to the real price of energy is assumed to be negative, i.e. $\chi > 0$, unless otherwise noted.

The model can be closed, for positive analysis purposes, by specifying an interest rate Taylor-type rule. This is assumed to take the following form,

$$\widetilde{I}_t = \phi_\pi \, \pi_{g,t} + \phi_f \, \widehat{f}_t \tag{44}$$

¹¹This reduced-form specification can be consistent with wage-setting schemes such as nominal wage indexation or Nash bargaining between workers and firm owners.

targeting core inflation as well as fluctuations in labour market slack, as captured by the job-finding rate, \hat{f}_t . In the model, $\pi_{g,t}$ is the welfare-relevant measure of inflation, as it is the one that is associated with price dispersion, reducing, in turn, the quantity of output that can be obtained from a given amount of inputs.¹² Blanchard and Galí (2007) similarly use a Taylor rule targeting core inflation, a variable they argue that many central banks appear to focus on as the basis for their interest rate decisions.¹³

5.1 Energy Prices and Aggregate Demand

Focusing on the demand block of the model, as captured by the dynamic IS equation, the percentage deviation of the gross real interest rate from its steady-state level, $\tilde{R}_t := \tilde{I}_t - \mathbb{E}_t(\pi_{t+1})$, can be expressed as follows, in first-order approximation,

$$\widetilde{R}_{t} \simeq \underbrace{-\frac{\Lambda}{\lambda} \mathbb{E}_{t}(\widehat{\lambda}_{t+1})}_{\text{Unemployment Risk}} + \underbrace{\frac{\chi(1-\rho_{e})\widetilde{p}_{e,t}}_{\text{Standard}}}_{\text{Consumption Smoothing}} + \underbrace{\frac{\Xi_{w}}{1-\Xi_{w}}(1+\chi)(1-\rho_{e})\widetilde{p}_{e,t}}_{\text{Non-homothetic}} \underbrace{-\Lambda\Psi\frac{\Xi_{w}}{1-\Xi_{w}}(1+\chi)\rho_{e}\widetilde{p}_{e,t}}_{\text{Heterogeneous Exposure}}$$
(45)

First of all, we can notice that in the perfect insurance limit, $\zeta \to 0$, we have $\Lambda = 0$, implying

$$\lim_{\zeta \to 0} \widetilde{R}_t \simeq \chi \left(1 - \rho_e\right) \widetilde{p}_{e,t} + \frac{\Xi_w}{1 - \Xi_w} \left(1 + \chi\right) \left(1 - \rho_e\right) \widetilde{p}_{e,t}$$

hence, in this case, aggregate demand is always sustained by an increase in real energy prices, by both a standard consumption smoothing channel and the non-homotheticity of consumer preferences. The former channel induces a desire to borrow as future wages are expected to bounce back up. The latter channel induces a desire to borrow as well, in order to smooth consumption as future energy prices —and the consumption losses they directly entail due to non-homothetic preferences are expected to be lower than in the current period.

In the presence of imperfect insurance, i.e. $\zeta, \Lambda \in (0, 1)$, we can first notice that when preferences are homothetic, and hence $\Xi_w = 0$, energy prices do not have any direct impact on the demand

 $^{^{12}}$ Similarly, in Bodenstein et al. (2008), it is core inflation that matters for welfare since core prices are sticky while the price of energy is assumed to be completely flexible.

¹³In line with the ECB's medium-term orientation in pursuing its primary objective of price stability, core inflation can serve as a valuable indicator for understanding medium-term developments in headline inflation. In the present setting, given a high persistence of energy price shocks, π_t will settle to $\pi_{g,t}$ in the periods after the shock.

side of the model as captured by equation (45), which therefore reduces to the same considered by Challe (2020),

$$\widetilde{R}_t \simeq -\frac{\Lambda}{\lambda} \mathbb{E}_t(\widehat{\lambda}_{t+1}) + \mathbb{E}_t(\widetilde{w}_{t+1} - \widetilde{w}_t).$$

Still, even if energy prices do not directly affect aggregate demand, they can *indirectly* affect it through unemployment risk and real wage fluctuations.

Non-homothetic preferences imply, along with the presence of imperfect unemployment insurance, that employed and unemployed workers are heterogeneously exposed to energy price shocks. This channel, captured by the last term in (45), induces a precautionary saving desire that depresses aggregate demand when real energy prices increase. Indeed, the consumption losses upon unemployment are increased as the shock weighs more on the unemployed, causing, in turn, an incentive to precautionarily save on the part of currently employed workers out of the fear of becoming unemployed and hence more exposed to the surge in energy prices.

Therefore, the net effect of non-homotheticity on aggregate demand depends on which channel prevails between the desire to borrow conditional on remaining employed, captured by the third term in (45), and the desire to save out of the fear of becoming unemployed and hence more exposed to the shock. One can easily check that the latter channel prevails when the shock is sufficiently persistent, namely when $\rho_e > 1 / (1 + \Lambda \Psi)$.

Assuming that a shock to the price of energy is quite persistent is not unreasonable. For instance, Blanchard and Galí (2007) argue that the real price of oil would be better characterised as non-stationary. In the present setting, when $\rho_e \rightarrow 1$, the *direct* impact of energy price increases on aggregate demand would undoubtedly be negative and amount to

$$-\Lambda \Psi = -\frac{\rho \left(1-f\right) \zeta}{\rho \left(1-f\right) \zeta + \left(1-\zeta - \Xi_w\right)} \left(\frac{1-\Xi_w}{1-\zeta - \Xi_w}\right)$$

which is larger, in absolute terms, the larger the proportional loss in income upon unemployment, ζ , as energy price increases end up weighing even more on the unemployed relative to the employed, and hence employed workers have more incentive to precautionarily save in direct response to these shocks.

5.2 Positive Analysis: an Illustrative Case

While a quantitative assessment of the behaviour of the linearised model will be performed later in Section 6 in comparison to normative analysis, to understand the various forces at work, it is useful to consider the following illustrative case, which allows to derive a simple and intuitive analytical solution for the unemployment response to energy price shocks.

I first assume constant real wages, implying in turn that only the two precautionary motives highlighted in equation (45) will be at work in determining saving desires. I also assume full worker reallocation ($\rho = 1$) implying $\hat{n}_t = \hat{f}_t$, and set the matching function parameter to $\alpha = 0.5$. Lastly, the policy rule parameters are set to $\phi_{\pi} = 1/\beta$, $\phi_f = 0$.

Under these assumptions, the first-order behaviour of the model is characterised by the following equations

$$\begin{split} \widehat{mc}_{g,t} &= -\frac{2\,\kappa}{p_g}\,\widehat{u}_t + \left[\frac{p_e}{p_g}\,\left(\frac{\gamma_e}{1-\gamma_e}\right) + \left(\frac{1}{1-\gamma_e}\right)\,\frac{\omega_e}{1-\omega_e}\right]\,\widetilde{p}_{e,t} \\ \pi_{g,t} &= \beta\,\mathbb{E}_t(\pi_{g,t+1}) + \Theta\,\widehat{mc}_{g,t} \\ \pi_t &= \pi_{g,t} + \frac{\omega_e}{1-\omega_e}\,(\widetilde{p}_{e,t} - \widetilde{p}_{e,t-1}) \\ \frac{1}{\beta}\,\pi_{g,t} - \mathbb{E}_t(\pi_{t+1}) &= -\frac{\Lambda}{1-f}\,\mathbb{E}_t(\widehat{u}_{t+1}) - \Xi_w\,\frac{(1+\Lambda\,\Psi)\,\mathbb{E}_t(\widetilde{p}_{e,t+1}) - \widetilde{p}_{e,t}}{1-\Xi_w} \end{split}$$

Combining the last three equations, the dynamic IS curve becomes

$$\frac{\Theta}{\beta} \,\widehat{mc}_{g,t} + (1-\rho_e) \,\frac{\omega_e}{1-\omega_e} \,\widetilde{p}_{e,t} = -\frac{\Lambda}{1-f} \,\mathbb{E}_t(\widehat{u}_{t+1}) + (1-\rho_e) \,\frac{\Xi_w}{1-\Xi_w} \,\widetilde{p}_{e,t} - \rho_e \,\frac{\Xi_w}{1-\Xi_w} \,\Lambda \,\Psi \,\widetilde{p}_{e,t} \,.$$

When $\rho_e \rightarrow 1$, this can be approximated by the following linear expectational difference equation

$$\widehat{u}_t = \frac{p_g}{2\kappa} \mathcal{B}_n \mathbb{E}_t(\widehat{u}_{t+1}) + \frac{p_g}{2\kappa} \mathcal{C}_e \, \widetilde{p}_{e,t} \tag{46}$$

where $\mathcal{B}_n = \frac{\beta}{\Theta} \left(\frac{\Lambda}{1-f}\right)$, $\mathcal{C}_e = \frac{p_e}{p_g} \left(\frac{\gamma_e}{1-\gamma_e}\right) + \left(\frac{1}{1-\gamma_e}\right) \frac{\omega_e}{1-\omega_e} + \Lambda \Psi \frac{\beta}{\Theta} \left(\frac{\Xi_w}{1-\Xi_w}\right)$, and local determinacy requires $\frac{p_g}{2\kappa} \mathcal{B}_n < 1$.

5.2.1 Perfect Insurance Limit

In the perfect insurance limit where the income of the unemployed is equal to that of the employed, the first-order response of unemployment to an increase in real energy prices is given by

$$\lim_{\zeta \to 0} \widehat{u}_t = \mathcal{F}_e \widetilde{p}_{e,t} \tag{47}$$

where $\mathcal{F}_e = \frac{p_g}{2\kappa} \left[\frac{p_e}{p_g} \left(\frac{\gamma_e}{1-\gamma_e} \right) + \left(\frac{1}{1-\gamma_e} \right) \frac{\omega_e}{1-\omega_e} \right].$

It is worth noticing that the perfect insurance limit corresponds, in the particular case considered here, to the natural (flexible-price) response, as the two precautionary motives shaping the response of the natural rate vanish in the perfect insurance limit, implying, in turn, a steady natural rate when real wages are assumed to remain constant. This perfect-insurance (and natural) response is, then, entirely driven by what happens on the producer side of the model: in response to rising real energy prices, firms demand less energy; given that this input is a complement with labour in production, and that real wages are assumed to remain fixed, also labour demand decreases, and hence unemployment rises at equilibrium.

5.2.2 Imperfect Insurance

Coming to the imperfect insurance case, and considering first a homothetic preferences situation, we have

$$\widehat{u}_t = \frac{\mathcal{F}_e}{1 - \frac{p_g}{2\kappa} \frac{\beta}{\Theta} \left(\frac{\zeta}{1 - \zeta f}\right)} \widetilde{p}_{e,t}$$
(48)

where the denominator shall be positive under local determinacy. The presence of imperfect unemployment insurance then amplifies the increase in unemployment in response to a rise in real energy prices: this amplification channel is captured by the presence of the second term in the denominator of equation (48). Intuitively, the increase in unemployment risk that follows the reduction in activity induces a precautionary saving desire on the part of households, which depresses aggregate demand. Due to price rigidity, this drag on aggregate demand ends up increasing unemployment at equilibrium. Moreover, the magnitude of the response is larger the larger the degree of imperfect insurance, as captured by the proportional income loss upon unemployment, ζ . In the non-homothetic case, we have

$$\widehat{u}_{t} = \frac{\mathcal{F}_{e} + \frac{p_{g}}{2\kappa} \frac{\beta}{\Theta} \left(\frac{\Xi_{w}}{1-\zeta-\Xi_{w}}\right) \left[\frac{(1-f)\zeta}{1-\zeta f-\Xi_{w}}\right]}{1 - \frac{p_{g}}{2\kappa} \frac{\beta}{\Theta} \left(\frac{\zeta f}{1-\zeta f-\Xi_{w}}\right)} \widetilde{p}_{e,t}$$
(49)

where, again, the denominator shall be positive under local determinacy. The presence of heterogeneous exposure to the shock due to non-homotheticity then implies an additional source of amplification of the unemployment response to a rise in real energy prices. First, as before, greater unemployment risk depresses aggregate demand and employment. This channel is again captured by the second term in the denominator of equation (49). Second, due to the presence of non-homothetic preferences, as captured by $\Xi_w \in (0, 1)$, the rise in real energy prices increases the consumption losses upon unemployment, thereby strengthening the drag on aggregate demand. This latter amplification channel is captured by the second term in the numerator of equation (49). As before, when ζ increases, the magnitude of the response increases as well, now through both the two channels.

6 Normative Analysis

In this section, I investigate the sources of the trade-off between stabilising core inflation and unemployment that arise in this economy. I first focus, in the spirit of Challe (2020), on a set of parametric restrictions that delivers a closed-form expression for the core inflation and unemployment rates under the optimal policy. This particular case with a closed-form analytical solution helps to gather the intuition about the joint role of imperfect unemployment insurance and non-homotheticity in determining a heterogeneous exposure to the shock, which stands as the primary source of the trade-off. The latter materialises as an endogenous wedge between natural unemployment (the level consistent with stable core inflation) and constrained-efficient unemployment. As a consequence, the monetary authority will not be able to replicate the constrained-efficient allocation by aiming to stabilise core inflation pressures and will need instead to seek a compromise between stabilising core inflation and unemployment at the constrained-efficient level.

I then move to a more general case, where the parameters will be disciplined from the data to calibrate the model and get a sense of its quantitative properties under numerical simulations.

6.1 Optimal Policy: an Illustrative Analytical Case

Assuming $\rho = 1$, we have $n_t = f_t = v_t^{1-\alpha}$. Above all, this assumption implies that employment does not act as a state variable in the welfare objective in (37), allowing, in turn, a full analytical treatment of the optimal policy problem. In this respect, also assuming $\alpha = 0.5$, the actual, natural, and constrained-efficient unemployment levels read, respectively,

$$u_t = 1 - \frac{1}{2\kappa} \left(\varphi_t - w_t + S\right) \tag{50}$$

$$u_t^n = 1 - \frac{1}{2\kappa} \left[\frac{p_{g,t} - \gamma_e \, p_{e,t}}{1 - \gamma_e} - w_t + S \right]$$
(51)

$$u_t^* = 1 - \frac{1}{2\kappa} \left[\frac{p_{g,t} - \gamma_e \, p_{e,t}}{1 - \gamma_e} - w_t^* + \frac{\ln(w_t^* - \xi \, p_{e,t}) - \ln(\delta_t - \xi \, p_{e,t})}{\nu} \right]$$
(52)

In terms of level deviations from the constrained-efficient steady state, we have, in first-order approximations,

$$\widehat{u}_t^n \simeq \mathcal{F}_e \, \widetilde{p}_{e,t} + \frac{w}{2\,\kappa} \, \widetilde{w}_t \tag{53}$$

$$\widehat{u}_t^* \simeq \mathcal{F}_e \, \widetilde{p}_{e,t} + \frac{w}{2\,\kappa} \, \widetilde{w}_t^* + \frac{w}{2\,\kappa} \, \Psi \, \left[(1-\zeta) \, \widetilde{\delta}_t - \Xi_w \, \widetilde{p}_{e,t} \right] \tag{54}$$

How differently the natural and constrained-efficient unemployment levels respond to energy price shocks then depends on how the decentralised wage w_t compares to the constrained-efficient wage $w_t^* = \frac{1}{\nu} + \xi p_{e,t}$, and on the cyclical behaviour of δ_t . In what follows, I assume again, as in Challe (2020) and as specified in (42), that the decentralised wage and home production fluctuate in such a way that the income loss upon unemployment is constant, implying, in turn, $\tilde{\delta}_t = \tilde{w}_t$, while the decentralised wage is assumed to evolve as in (43), i.e. $\tilde{w}_t = -\chi \tilde{p}_{e,t}$, with $\chi > 0$ unless otherwise noted. Therefore, the wedge between natural and constrained-efficient unemployment is

$$\widehat{u}_t^n - \widehat{u}_t^* = \frac{w}{2\kappa} \zeta \Psi \left(\frac{\Xi_w}{1 - \Xi_w}\right) (1 + \chi) \widetilde{p}_{e,t}.$$
(55)

Now, given the New Keynesian Phillips Curve (NKPC)

$$\pi_{g,t} = \beta \mathbb{E}_t(\pi_{g,t+1}) + \Theta \,\widehat{mc}_{g,t}$$

under the simplifying assumptions made in this section, we have $\widehat{mc}_{g,t} = -\frac{2\kappa}{p_g} (\widehat{u}_t - \widehat{u}_t^n)$. Letting

 $\mathcal{U}_t := \widehat{u}_t - \widehat{u}_t^*$ denote the welfare-relevant unemployment gap,

$$\pi_{g,t} = \beta \mathbb{E}_t(\pi_{g,t+1}) - \Theta \frac{2\kappa}{p_g} \mathcal{U}_t + \underbrace{\Theta \frac{w}{p_g} \zeta \Psi\left(\frac{\Xi_w}{1 - \Xi_w}\right) (1 + \chi) \widetilde{p}_{e,t}}_{\text{cost-push term}}$$
(56)

and we can see that energy price shocks, by inducing a time-varying wedge between natural and constrained-efficient unemployment, act endogenously as a cost-push term in the NKPC. I now turn to discuss its primary sources.

Heterogeneous Exposure to Energy Price Shocks. If the constrained-efficient wage could be decentralised also outside the steady state, i.e. $\chi = -\Xi_w$, there would still be a wedge between natural and constrained-efficient unemployment, amounting to

$$\widehat{u}_t^n - \widehat{u}_t^* = \frac{w}{2\kappa} \zeta \, \Psi \, \Xi_w \, \widetilde{p}_{e,t}$$

In other words, a wedge between natural and constrained-efficient unemployment arises endogenously, first of all, due to the presence of subsistence energy consumption and imperfect unemployment insurance, which jointly imply that the unemployed are more exposed to energy price shocks than the employed. Indeed, as a result, rising energy prices induce an increase in the consumption losses upon unemployment. These would be efficiently counteracted by larger employment subsidies, but as these remain constant at their steady-state amount S, a wedge between natural and constrained-efficient unemployment arises.

This, then, constitutes a novel source of a cost-push term in the NKPC compared to the benchmark setting of Challe (2020). Indeed, when $\Xi_w = 0$ and hence preferences are homothetic, we can see that the cost-push term vanishes. The term vanishes also in the perfect insurance limit $\zeta \to 0$, as in both these cases there is no heterogeneity between the employed and the unemployed in their direct exposure to energy price shocks.

Real Wage Rigidity. While in Challe (2020) constant real wages can be constrained-efficient (when they correspond to their constrained-efficient steady-state amount) here, as already mentioned, due to the presence of subsistence energy needs, a constant real wage rate cannot insure workers against fluctuations in their consumption bundle. As a result, real wage rigidity constitutes another novel source of endogenous cost-push terms, in contrast with Challe (2020). In particular, we can see that when constrained-efficient wage fluctuations cannot be decentralised and real wages remain

constant (i.e. $\chi = 0$), the wedge becomes of larger magnitude

$$\widehat{u}_t^n - \widehat{u}_t^* = \frac{w}{2 \kappa} \zeta \Psi \left(\frac{\Xi_w}{1 - \Xi_w} \right) \, \widetilde{p}_{e,t} \, .$$

Compared to the constant real wage case, on the one hand the downward adjustment (captured by $\chi > 0$) moderates the rise in natural unemployment. However, on the other hand, it also causes the home production of the unemployed to fall, calling for a larger employment subsidy on the constrained-efficient side. This latter effect dominates, driving a wedge between natural and constrained-efficient unemployment of larger magnitude compared to the case of constant real wages, as can be seen from (55).

6.1.1 Linear-quadratic Problem

In this setting, similarly to Challe (2020), one can derive the following expression for welfare losses from a second-order approximation of the welfare $objective^{14}$

$$L_t = \frac{1}{2} \mathbb{E}_t \sum_{j=0}^{\infty} \beta^j \left(\mathcal{U}_{t+j}^2 + \Omega \, \pi_{g,t+j}^2 \right)$$
(57)

where $\Omega = \frac{n \varepsilon}{(1-\gamma_e)} \frac{p_g}{2\kappa} / \Theta$. The loss function in (57) makes explicit that $\pi_{g,t}$ is the welfare-relevant measure of inflation in this economy, as mentioned before.

Optimal Discretionary Policy. The optimal discretionary policy minimises the period losses

$$\frac{1}{2} \left(\mathcal{U}_t^2 + \Omega \, \pi_{g,t}^2 \right)$$

subject to (56) taking expected inflation as given. The solution to the problem (detailed in Appendix B.6) gives the following targets for core inflation and the welfare-relevant unemployment gap

$$\pi_{g,t} = \Upsilon \zeta \Psi \left(\frac{\Xi_w}{1 - \Xi_w}\right) (1 + \chi) \widetilde{p}_{e,t}$$
(58)

$$\mathcal{U}_t = \left(\frac{n\,\varepsilon}{1-\gamma_e}\right)\,\Upsilon\,\zeta\,\Psi\,\left(\frac{\Xi_w}{1-\Xi_w}\right)\,(1+\chi)\,\widetilde{p}_{e,t} \tag{59}$$

where $\Upsilon = \Theta \frac{w}{p_g} \left(1 - \beta + \Theta \frac{2\kappa}{p_g} \frac{n\varepsilon}{(1-\gamma_e)} \right)^{-1}$. We can then see that, in response to an increase in real energy prices, some core inflation is optimally accommodated in order to achieve a smaller gap

¹⁴See Appendix B.5 for more details.

between actual and constrained-efficient unemployment. Indeed, one can easily check that if core inflation were fully stabilised, the welfare-relevant unemployment gap would amount to

$$\mathcal{U}_t = \frac{w}{2\kappa} \zeta \Psi \left(\frac{\Xi_w}{1 - \Xi_w}\right) (1 + \chi) \widetilde{p}_{e,t}$$

therefore implying higher unemployment as $\frac{w}{2\kappa} > \left(\frac{n\varepsilon}{1-\gamma_e}\right) \Upsilon$.

Intuitively, the optimal discretionary policy involves partly accommodating core inflation in order to indirectly sustain employment and, hence, avoid too many households flowing to unemployment, thereby becoming more exposed to the shock. We can also notice that when preferences are homothetic (corresponding to $\Xi_w = 0$), and hence the employed and the unemployed are homogeneously exposed to the shock, the trade-off vanishes, and the monetary authority can optimally stabilise core inflation and the welfare-relevant unemployment gap simultaneously.

Lastly, we can see that the targets imply, at equilibrium, that the nominal interest rate consistent with the optimal discretionary policy satisfies

$$\widetilde{I}_{t} \simeq \widetilde{R}_{t}^{n} + \frac{\Lambda}{1-f} \underbrace{\left(\frac{1-\beta}{\Theta}\right) \frac{p_{g}}{2\kappa} \Upsilon \zeta \Psi \left(\frac{\Xi_{w}}{1-\Xi_{w}}\right) (1+\chi) \widetilde{p}_{e,t}}_{\mathbb{E}_{t}(\widetilde{x}_{t+1}^{n})} + \underbrace{\Upsilon \zeta \Psi \left(\frac{\Xi_{w}}{1-\Xi_{w}}\right) (1+\chi) \widetilde{p}_{e,t}}_{\mathbb{E}_{t}(\pi_{g,t+1})}$$
(60)

where we can notice that in the absence of a heterogeneous exposure to energy price shocks, when $\Xi_w = 0$, the optimal policy *outcome* is to align the nominal rate to the real natural rate, thereby sterilising core inflation pressures, just as in the textbook Representative-Agent New Keynesian (RANK) model. By contrast, when unemployed and employed households are heterogeneously exposed to these shocks, a wedge between natural and constrained-efficient unemployment arises endogenously, and the optimal policy prescription changes. From (60), we can see that the equilibrium real interest rate ends up being higher than the natural real rate, $\tilde{I}_t - \mathbb{E}_t(\pi_{t+1}) > \tilde{R}_t^n$: since unemployment is lower than its natural level, equilibrium demand is kept afloat as the precautionary saving motive due to unemployment risk is dampened.

Optimal Ramsey Policy. The optimal Ramsey policy minimises the lifetime welfare losses in (57) subject to the sequence of constraints posed by (56), giving the following targeted *paths* (formally derived in Appendix B.7) for core inflation and the welfare-relevant unemployment gap

$$\pi_{g,t+j} = \eta^j \,\overline{\Upsilon} \,\zeta \,\Psi \,\left(\frac{\Xi_w}{1-\Xi_w}\right) \,(1+\chi) \,\widetilde{p}_{e,t} \tag{61}$$

$$\mathcal{U}_{t+j} = \left(\frac{n\,\varepsilon}{1-\gamma_e}\right) \, \left(\frac{1-\eta^{j+1}}{1-\eta}\right) \, \overline{\Upsilon} \, \zeta \, \Psi \, \left(\frac{\Xi_w}{1-\Xi_w}\right) \, (1+\chi) \, \widetilde{p}_{e,t} \tag{62}$$

for $j \ge 0$, where $\eta := \frac{1+\beta+\Theta\frac{2\kappa}{p_g}\frac{n\varepsilon}{(1-\gamma_e)}}{2\beta} \left[1 - \sqrt{1 - 4\beta\left(1 + \beta + \Theta\frac{2\kappa}{p_g}\frac{n\varepsilon}{(1-\gamma_e)}\right)^{-2}}\right] \in (0,1)$ and $\overline{\Upsilon} = \Theta \frac{w}{p_g} \left(1 - \eta \beta + \Theta \frac{2\kappa}{p_g} \frac{n\varepsilon}{(1 - \gamma_e)} \right)^{-1} < \Upsilon$. Therefore, just as in the textbook RANK model, if the monetary authority is able to credibly commit to this policy plan, it can achieve a smaller welfare-relevant unemployment gap on impact and a smaller increase in core inflation relative to the optimal discretionary policy. The main intuition is, however, the same as in the discretionary case: core inflation is partly accommodated in order to indirectly sustain employment, thereby preventing too many workers from ending up being more exposed to the shock by becoming unemployed.

6.2 **Optimal Policy:** the General Case

This section considers the general case of $\rho \in (0, 1)$, with the aim of giving a sense of the quantitative properties of the model. However, it should be stressed that the quantitative predictions of the model, given its stylised nature, should not be taken too literally.

When $\rho < 1$, employment acts as a state variable in the welfare objective since the existing employment stock impacts the current amount of hiring costs, as can be seen from (34). This requires resorting to a numerical solution of the Ramsey problem,¹⁵ and hence choosing first of all an appropriate parametrisation for the model.

6.2.1Parametrisation

The parametrisation, summarised in Table 2, is aimed to target an average continental European labour market, with each time period taken to correspond to a quarter.

Preferences. Given a quarterly frequency, the discount factor is set to $\beta = 0.98$, below its usual value of 0.99. Indeed, the presence of uninsured unemployment risk brings the average real rate of return below the value of $1/\beta$ that would prevail under complete markets.¹⁶ Hence, β must be calibrated to a lower value to ensure that real rates of return are not unrealistically small.

The parameters governing the relative weight of energy and the degree of non-homotheticity in consumer preferences (ω_e and ξ , respectively) are calibrated by exploiting the following results

¹⁵The numerical solution is obtained using Dynare (www.dynare.org). ¹⁶Given the steady state Euler condition $1 = \beta R \left[(1 - \lambda) + \lambda \left(\frac{w - p_e \xi}{\delta - p_e \xi} \right) \right]$, one can easily see that since $\left[(1-\lambda)+\lambda\,\left(\tfrac{w-p_e\,\xi}{\delta-p_e\,\xi}\right)\right]>1,\,\text{then }R<1/\beta.$

(formally derived in Appendix B.2)

$$\omega_e^n := \frac{p_e e^n}{w} = \omega_e + (1 - \omega_e) \Xi_w \tag{63}$$

$$\omega_e^u := \frac{p_e \, e^u}{\delta} = \omega_e + (1 - \omega_e) \, \frac{\Xi_w}{1 - \zeta} \tag{64}$$

along with the empirical evidence from Table 1 on the energy shares of the employed and the unemployed —appearing on the LHS of equations (63) and (64)— and on the average percentage difference in overall consumption expenditure between them, which is accordingly set to $\zeta = 22\%$. Reassuringly, this is close to the 20% consumption loss *upon* unemployment documented by Chodorow-Reich and Karabarbounis (2016) and in the range of the estimates surveyed by Den Haan et al. (2018). Therefore, (63) and (64) jointly imply $\Xi_w = 0.11$ and $\omega_e = 0.08$, with the latter ending up being in line with the weight of energy in the euro-area Harmonised Index of Consumer Prices. **Labour Market.** Following Blanchard and Galí (2010), I target an average job-finding rate f = 0.25 and an average unemployment rate u = 0.10, implying, in turn, a separation rate $\rho = 0.037$.

The matching function elasticity is set to $\alpha = 0.6$, which is the midpoint of the range of estimates surveyed by Petrongolo and Pissarides (2001). As in Challe (2020), flow vacancy costs κ are taken to be 4.5% of the real wage rate. Given the targets, the latter can be recovered from (41), and its negative elasticity with respect to the real price of energy is set to $\chi = 0.1$, which would imply a decline in real wages of 4% in light of a 40% increase in the real price of energy, roughly in line with what has been observed in the euro area during the current energy crisis.¹⁷

Producers. The energy share in production is set to $\gamma_e = 0.04$, as in Bachmann et al. (2022) and Pieroni (2023). The Calvo pricing parameter is set to $\theta = 3/4$, implying an average price duration of one year. The elasticity of substitution among core good varieties is set to $\varepsilon = 4$, as in Gagliardone and Gertler (2023), which would imply a steady-state markup of 1/3;¹⁸ even when this is offset by the subsidy $\tau_y = 1/\varepsilon$, the parameter ε might still influence the relative welfare weight: for instance, a smaller ε implies less weight on inflation in the quadratic welfare loss function in (57).

Energy Price Shock. I set the shock as a 40% increase in the real price of energy,¹⁹ hitting the economy with high persistence; namely, $\rho_e = 0.97$ as in Blanchard and Galí (2007), who choose their autoregressive parameter to have the real price of oil very close to a random walk, as it is in the data, while retaining stationarity.

¹⁷I check that the real wage remains always in the bargaining set along the simulated path.

¹⁸Kouvavas et al. (2021) document an average markup for the euro area of around 0.45 over the 1995—2018 period. ¹⁹This aligns with the recent peak rise of the eurozone real energy price index above its long-run average in 2022 Q4.

Targets			
	Description	Value	Source
f	Average job-finding rate	0.25	Blanchard and Galí (2010)
u	Average unemployment rate	0.10	Blanchard and Galí (2010)
κ/w	Flow vacancy cost	0.045	Challe (2020)
ζ	Expenditure loss upon unemployment	0.22	ECB Consumer Expectation Survey
ω_e^n	Energy share of the employed	0.182	ECB Consumer Expectation Survey
ω_e^u	Energy share of the unemployed	0.211	ECB Consumer Expectation Survey

Table 2: PARAMETRISATION

Parameters			
	Description	Value	Source/Targets
β	Discount factor	0.98	4% annual interest rate
ω_e	Quasi-share of energy in consumption	0.08	$\zeta, \omega_e^n, \omega_e^u$
ξ	Subsistence consumption of energy	0.109	$\zeta, \omega_e^n, \omega_e^u$
ho	Separation rate	0.037	f, u
α	Matching function elasticity	0.6	Petrongolo and Pissarides (2001)
θ	Fraction of unchanged prices	0.75	1 year avg. price duration
ε	Elasticity of Substitution	4	1/3 steady-state markup
γ_e	Share of energy in production	0.05	Bachmann et al. (2022)
$ ho_e$	Energy price shock persistence	0.97	Blanchard and Galí (2007)

Even if the increase in the energy share of the unemployed relative to that of the employed is not explicitly targeted, the chosen parametrisation is able to match the one percentage point documented in the motivating evidence summarised in Figure 2. Indeed, given the energy shares formally derived in Appendix B.2, when the shock hits $\widehat{\omega}_{e,0}^u - \widehat{\omega}_{e,0}^n \simeq (\omega_e^u - \omega_e^n) (1 + \chi) \widetilde{p}_{e,0} = 0.01$. This is reassuring, as the increased relative exposure to the shock on the part of the unemployed is suggested to be the key driver of the monetary policy trade-off uncovered analytically in Section 6.1, and the proposed parametrisation is able to quantitatively match this magnitude with the evidence.

6.2.2 Impulse Responses

Figure 3 shows the responses of core inflation and unemployment to a one-off increase in the real price of energy, comparing the behaviour of the linearised model under the optimal Ramsey policy with that under two alternative policy rules of the type of equation (44): a strict inflation targeting

rule (with $\phi_{\pi} = 15$, $\phi_f = 0$), aimed at suppressing core inflationary pressures, and a rule aimed at stabilising the unemployment rate by also targeting fluctuations in labour market slack and reacting to inflation less aggressively (with $\phi_{\pi} = 1.5$, $\phi_f = 1.5$). For all these cases, I consider a scenario where the employed and the unemployed are heterogeneously exposed to the shock, as it is suggested by the evidence from Section 2, and a counterfactual scenario where instead preferences are homothetic (i.e. $\xi = \Xi_w = 0$) and hence the employed and the unemployed are homogeneous in their direct exposure to the shock. The top panel of Figure 3 reports the impulse responses from the former case, while the bottom panel those from the latter case.

In the heterogeneous exposure case, the surge in the real price of energy causes a sizeable rise in unemployment under strict inflation targeting. By contrast, under the optimal Ramsey policy, the monetary authority is able to achieve a notably smaller increase in unemployment at the cost of partial accommodation of core inflation, as suggested by the analytical results. Again, this is intuitively optimal as, in such case, fewer workers lose their jobs and hence avoid becoming more exposed to the surge in energy prices. However, a full stabilisation of the unemployment rate is also not optimal, as it implies an overshooting of core inflation compared to the optimal policy: even if unemployment barely increases on impact under the latter, a smaller rise in inflation is achieved by credibly committing to a future recession so as to dampen expectations about future price rises.

In the homogeneous exposure case, as also was suggested by the analytical results, the trade-off disappears, and the optimal Ramsey policy coincides with that achieved under a Taylor rule aimed at aggressively targeting core inflation.

It is also worth mentioning that the results would be substantially unchanged, over the medium term, if overall inflation instead of core inflation were the target of the Taylor rule: indeed, since $\pi_t = \pi_{g,t} + \frac{\omega_e}{1-\omega_e} (\tilde{p}_{e,t} - \tilde{p}_{e,t-1})$, and ρ_e is large while ω_e is relatively small, we have that with the exception of the period when the shock hits, in subsequent periods $\frac{\omega_e}{1-\omega_e} (\tilde{p}_{e,t} - \tilde{p}_{e,t-1}) \simeq 0$ and $\pi_t \simeq \pi_{g,t}$. However, if overall inflation were to be fully stabilised also in the period of the shock, this would require core deflation, implying a larger surge in unemployment than optimal also in the homogeneous exposure case, where instead stabilising only core inflation is optimal. In other words, in this case, the optimal policy is still accommodative towards the energy component of consumer inflation but hard on core inflation (similarly to Natal, 2012), as can be seen in Figure 4.

Taking stock, even if the model is highly stylised and its exact quantitative predictions should not be taken too literally, its qualitative predictions are likely to matter in practice and are in line with those suggested by the analytical results: compared to a policy rule aimed at fully stabilising

Figure 3: UNEMPLOYMENT AND INFLATION EFFECTS OF AN ENERGY PRICE SHOCK



(a) Heterogeneous Exposure to the Shock

(b) Homogeneous Exposure to the Shock



Notes: The figure reports model-based impulse responses to a 40% real energy price shock of persistence $\rho_e = 0.97$ (left panels), under the parametrisation summarised in Table 2. Solid lines correspond to responses under the Ramsey-optimal monetary policy, dotted lines to those under strict inflation targeting ($\phi_{\pi} = 15, \phi_f = 0$), and dashed lines to those under a stable unemployment policy ($\phi_{\pi} = 1.5, \phi_f = 1.5$). Middle and right panels report, respectively, the responses of core inflation and unemployment under the alternative monetary policy regimes. The top panels refer to the heterogeneous exposure case ($\Xi_w = 0.11$) while the bottom panels to the homogeneous exposure case ($\xi = \Xi_w = 0$, other parameters unchanged).



Figure 4: Overall Inflation and its Components Under the Optimal Ramsey Policy

Notes: The figure reports model-based impulse responses, in the Ramsey-optimal monetary policy regime, to a 40% real energy price shock of persistence $\rho_e = 0.97$, under the parametrisation summarised in Table 2. Solid lines correspond to overall inflation, π_t , red areas to its energy component, $\omega_e \pi_{e,t}$, and green areas to its core (non-energy) component, $(1 - \omega_e) \pi_{g,t}$. The left panel refers to the heterogeneous exposure case ($\Xi_w = 0.11$) while the right panel to the homogeneous exposure case ($\xi = \Xi_w = 0$, other parameters unchanged).

core inflation, the optimal policy is able to achieve a smaller increase in unemployment at the cost of partly accommodating core inflation; the rise in inflation is, however, smaller than if unemployment instead were to be fully stabilised. This trade-off arises due to the heterogeneous exposure to the increase in energy prices between the employed and the unemployed: in the case of homogeneous direct exposure to the shock, the optimal policy coincides with one aimed at fully stabilising core inflation.

7 Conclusions

This paper provides novel evidence on the heterogeneous exposure to energy price shocks between the employed and the unemployed, studying the implications of this fact for the (optimal) conduct of monetary policy in an analytically tractable HANK-S&M model, where non-homothetic preferences and imperfect unemployment insurance endogenously give rise to the heterogeneous exposure documented from the data. Rising energy prices induce a novel precautionary saving motive in this setting, as the consumption losses upon unemployment are increased, posing a drag on aggregate demand and employment. Moreover, the shock acts endogenously as a cost-push term, implying that the monetary authority optimally accommodates some core inflation so as to contain the rise in unemployment and, hence, avoid households becoming more exposed to the shock.

Even if this paper analyses a specific dimension of heterogeneity, it can lend potentially broader insights: some accommodation of core inflation might be needed to prevent households from becoming poorer and, hence, more hit by energy price shocks. Also, these insights can easily extend to similar situations when other subsistence goods besides energy receive sizeable relative price shocks. The monetary policy trade-off highlighted in this work can likely only be dampened, but not eliminated, by transfers from employed to unemployed workers. Even if, for the sake of exposure, this aspect was not explicitly modelled, these transfers would still provide only partial insurance to the unemployed. Indeed, the fiscal authority would be faced with yet another trade-off, between dampening the increased consumption losses upon unemployment with more generous insurance on the one hand and, on the other hand, the fact that this would depress job search incentives. Lastly, this paper considers a zero-liquidity economy, an assumption which allows analytical and numerical tractability of the optimal monetary policy problem. Therefore, the results focus on the effect of energy price shocks on consumption inequality between the employed and the unemployed, regardless of the potential wealth redistribution effects of inflation and monetary policy across households.

Appendix

A Data Description

The ECB Consumer Expectations Survey (CES) is a panel survey of consumers that has been carried out on a monthly basis starting effectively in April 2020. Its goal is to provide timely, high-frequency information on the perceptions and expectations of euro-area consumers about the economy, as well as their economic and financial behaviour. The microdata are collected in the six main euro-area countries: Belgium, Germany, Spain, France, Italy, and the Netherlands.²⁰ The collected sample aims to be representative of the surveyed population by age, gender, and residence region. Table A.1 summarises the main variables of interest for the analysis of Section 2, which are described in more detail in this appendix section.

Expenditure on goods and services. On a quarterly basis, interviewed individuals are asked how much their household spent during the last month on goods and services as listed in twelve major categories. These are: (1) Food, beverages, groceries, tobacco; (2) Restaurants (including take-out food and delivery), cafes/ canteens; (3) Housing (including rent, maintenance/repair costs, home owner/renter insurance, but excluding mortgage payments); (4) Utilities (including water, sewerage, electricity, gas, heating oil, phone, cable, internet); (5) Furnishings, household equipment, small appliances and routine maintenance of the house; (6) Debt repayments (installments in mortgage, consumer loans, car loans, credit cards, student loans, other loans); (7) Clothing, footwear; (8) Health (including personal care products and services), health insurance; (9) Transport (fuel, car maintenance, public transportation fares); (10) Travel, recreation, entertainment and culture; (11) Childcare and education (including tuition fees for child and adult education, costs of after school activities, care of children/ babysitting, but excluding installments on student loans); (12) Other expenditures not mentioned above.

Employment situation. Starting from the October 2020 wave of the survey, on a quarterly basis interviewed individuals are asked what best describes their current employment situation and can give one of the following answers: (1) Working full-time (self-employed or working for someone else); (2) Working part-time (self-employed or working for someone else); (3) Temporarily laid-off (expecting to return to the previous workplace); (4) On extended leave (disability, sick, parental or other leave); (5) Unemployed and actively looking for a job; (6) Unemployed, interested in having a

 $^{^{20}\}mathrm{In}$ 2022, the sample was extended to cover five additional countries: Ireland, Greece, Austria, Portugal and Finland.

Expenditure on goods and ser	rvices — food		
Expenditure on goods and ser	vices — restaurants		
Expenditure on goods and ser	rvices — housing		
Expenditure on goods and ser	rvices — utilities		
Expenditure on goods and ser	rvices — furnishing		
Expenditure on goods and ser	vices — debt repayment		
Expenditure on goods and ser	rvices - clothing		
Expenditure on goods and ser	vices — health		
Expenditure on goods and services — transport services			
Expenditure on goods and services — travel and recreation			
Expenditure on goods and services — childcare and education			
Expenditure on goods and services — other			
 Employment situation:	employed/unemployed/inactive		
 Age at the start of the year:	18-34/35-49/50-70/71+		
Gender:	male/female/other		
Birth abroad:	yes/no		
Education:	low/middle/high		
Presence of a partner:	yes/no		

job but not actively looking for a job; (7) Unable to work because of disability or other medical reasons; (8) In retirement or early retirement; (9) Studying, at school, or in training; (10) Looking after children or other persons, doing housework; (11) Other.

These 11 categories above are regrouped by the ECB into the following groups: Employed, if working full- or part-time, or if temporarily laid-off or on extended leave. Unemployed, if actively looking for a job, or interested in having a job but not actively looking for it.²¹ Inactive, in the remaining cases.

Age at the start of the year. When recruited for the survey, individuals are asked about their birth month and year, and the derived age variable is updated at the beginning of each year based on the reported date of birth. The responses are then regrouped in the age brackets indicated in Table A.1.

 $^{^{21}}$ This category includes all respondents who consider themselves unemployed, including those who consider themselves unemployed but are not actively searching for a job. Thus, the definition of unemployment used by the ECB in the CES does not necessarily correspond to that of official EU labour market statistics.

Education. During recruitment, individuals are asked what is the highest level of school they have completed, or the highest degree they have received. Possible responses are: (1) Primary or no education; (2) Lower secondary education; (3) High school diploma (or equivalent professional degree); (4) Some college but no academic degree (for example: no BA, BS); (5) Bachelor's Degree (for example: BA, BS) or equivalent professional degree; (6) Master's Degree (for example: MA, MBA, MS, MSW) or equivalent; (7) Doctoral Degree (for example: PhD) or equivalent. These categories are then regrouped, before data dissemination, into three groups: low (no education, primary, or secondary), middle (high school diploma, or some college but no academic degree), and high (bachelor's or higher degree).

B Derivations and Proofs

B.1 Household Problem

Employed workers

$$V^{n}(B_{t-1}, p_{e,t}) = \max_{\{c_{t}^{n}, g_{t}^{n}, e_{t}^{n}, B_{t}\}} \{\ln(c_{t}^{n}) + \beta \mathbb{E}_{t} \left[(1 - \lambda_{t+1}) V^{n}(B_{t}, p_{e,t+1}) + \lambda_{t+1} V^{u}(B_{t}, p_{e,t+1}) \right] \}$$

s.t.
$$\begin{cases} P_{e,t} e_{t}^{n} + P_{g,t} g_{t}^{n} + B_{t} \leq W_{t} + (1 + i_{t-1}) B_{t-1} \\ c_{t}^{n} \leq (g_{t}^{n})^{(1 - \omega_{e})} (e_{t}^{n} - \xi)^{\omega_{e}} \\ B_{t} \geq 0 \end{cases}$$

Unemployed workers

$$\begin{split} V^u(B_{t-1},p_{e,t}) &= \max_{\{c_t^u,g_t^u,e_t^u,B_t\}} \left\{ \ln(c_t^u) + \beta \operatorname{\mathbb{E}}_t \left[f_{t+1} \, V^n(B_t,p_{e,t+1}) + (1-f_{t+1}) \, V^u(B_t,p_{e,t+1}) \right] \right\} \\ &\qquad \text{s.t.} \, \begin{cases} P_{e,t} \, e_t^u + P_{g,t} \, g_t^u + B_t \leq \Delta_t + (1+i_{t-1}) \, B_{t-1} \\ c_t^u \leq (g_t^u)^{(1-\omega_e)} (e_t^u - \xi)^{\omega_e} \\ B_t \geq 0 \end{cases} \end{split}$$

Letting Γ_t^i be the multiplier associated to the budget constraint of household $i \in \{n, u\}$, and μ_t^i that associated with the consumption aggregator, first-order conditions are

$$\begin{split} & \left[\frac{\partial}{\partial c_t^i}\right]: \quad \frac{1}{c_t^i} = \mu_t^i \\ & \left[\frac{\partial}{\partial g_t^i}\right]: \quad P_{g,t} \, \Gamma_t^i = (1 - \omega_e) \, \frac{c_t^i}{g_t^i} \, \mu_t^i \\ & \left[\frac{\partial}{\partial g_t^i}\right]: \quad P_{e,t} \, \Gamma_t^i = \omega_e \, \frac{c_t^i}{e_t^i - \xi} \, \mu_t^i \\ & \left[\frac{\partial}{\partial B_t}\right]: \quad \begin{cases} \Gamma_t^n \geq \beta \, \mathbb{E}_t \left[(1 - \lambda_{t+1}) \, \frac{\partial V^n(B_t, p_{e,t+1})}{\partial B_t} + \lambda_{t+1} \, \frac{\partial V^u(B_t, p_{e,t+1})}{\partial B_t}\right] \\ & \Gamma_t^u \geq \beta \, \mathbb{E}_t \left[f_{t+1} \, \frac{\partial V^n(B_t, p_{e,t+1})}{\partial B_t} + (1 - f_{t+1}) \, \frac{\partial V^u(B_t, p_{e,t+1})}{\partial B_t}\right] \end{cases} \end{split}$$

Combining the first three FOCs gives the following demand schedules

$$g_t^i = (1 - \omega_e) \frac{P_t}{P_{g,t}} c_t^i \tag{B.1}$$

$$e_t^i - \xi = \omega_e \, \frac{P_t}{P_{e,t}} \, c_t^i \tag{B.2}$$

where

$$P_t := \frac{\mu_t^i}{\Gamma_t^i} = \left(\frac{P_{g,t}}{1 - \omega_e}\right)^{(1 - \omega_e)} \left(\frac{P_{e,t}}{\omega_e}\right)^{\omega_e} . \tag{B.3}$$

The envelope conditions are

$$\frac{\partial V^i(B_{t-1}, p_{e,t})}{\partial B_{t-1}} = (1 + i_{t-1}) \Gamma_t^i$$

hence, since $\Gamma_t^i = 1/(P_t c_t^i)$, we get the following Euler conditions

$$\frac{1}{c_t^n} \ge \beta \mathbb{E}_t \left\{ \left(\frac{1+i_t}{1+\pi_{t+1}} \right) \left[(1-\lambda_{t+1}) \frac{1}{c_{t+1}^n} + \lambda_{t+1} \frac{1}{c_{t+1}^u} \right] \right\}$$
(B.4)

$$\frac{1}{c_t^u} \ge \beta \mathbb{E}_t \left\{ \left(\frac{1+i_t}{1+\pi_{t+1}} \right) \left[f_{t+1} \frac{1}{c_{t+1}^n} + (1-f_{t+1}) \frac{1}{c_{t+1}^u} \right] \right\}$$
(B.5)

Now, given the demand schedules for energy and non-energy goods, and the consumption aggregator and price index, total household expenditure is given by

$$X_t^i := P_{g,t} g_t^i + P_{e,t} e_t^i = P_t c_t^i + P_{e,t} \xi$$
(B.6)

giving in turn

$$c_t^i = \frac{X_t^i}{P_t} - \frac{P_{e,t}}{P_t}\,\xi$$

therefore, since at equilibrium $X^u_t = \Delta_t$ and $X^n_t = W_t$, Euler conditions become

$$\frac{1}{w_t - p_{e,t}\xi} \ge \beta \mathbb{E}_t \left\{ \left(\frac{1 + i_t}{1 + \pi_{t+1}} \right) \left[(1 - \lambda_{t+1}) \frac{1}{w_{t+1} - p_{e,t+1}\xi} + \lambda_{t+1} \frac{1}{\delta_t - p_{e,t+1}\xi} \right] \right\}$$
(B.7)

$$\frac{1}{\delta_t - p_{e,t}\xi} \ge \beta \mathbb{E}_t \left\{ \left(\frac{1 + i_t}{1 + \pi_{t+1}} \right) \left[f_{t+1} \frac{1}{w_{t+1} - p_{e,t+1}\xi} + (1 - f_{t+1}) \frac{1}{\delta_{t+1} - p_{e,t+1}\xi} \right] \right\}$$
(B.8)

where wage income, w_t , home production, δ_t , and the energy price, $p_{e,t}$, are expressed in real terms (i.e. in consumption basket units).

B.2 Implications of Non-homotheticity of Household Preferences

Own-price Elasticity of Energy Demand.

$$\varepsilon_{e,e} = \frac{\partial e}{\partial P_e} \frac{P_e}{e} = -\frac{\omega_e X}{\omega_e X + (1 - \omega_e) P_e \xi}$$
(B.9)

demand for energy is inelastic, as $|\varepsilon_{e,e}| < 1$.

Cross-price Elasticity of Demand for Non-energy Goods.

$$\varepsilon_{g,e} = \frac{\partial g}{\partial P_e} \frac{P_e}{g} = -(1 - \omega_e) \frac{P_e \xi}{P_g g}$$
(B.10)

since $\varepsilon_{g,e} < 0, g$ is a gross complement of e.

Income Elasticity of Relative Demand.

$$\varepsilon_{X,e/g} = \frac{\partial e/g}{\partial X} \frac{X}{e/g} = -\frac{P_e \xi}{X} \left(\frac{P_e \xi}{X}\right) \left(\frac{X}{X - P_e \xi}\right)^2 \tag{B.11}$$

therefore, as X falls, households consume more e relative to g.

Income Share of Energy. At equilibrium, since households spend all their income, we have

$$P_{g,t} g_t^u + P_{e,t} e_t^u = \Delta_t \tag{B.12}$$

$$P_{q,t}g_t^n + P_{e,t}e_t^n = W_t \tag{B.13}$$

therefore, given (B.6),

$$P_t c_t^u = \Delta_t - P_{e,t} \xi \tag{B.14}$$

$$P_t c_t^n = W_t - P_{e,t} \xi \tag{B.15}$$

hence,

$$P_{e,t} e_t^n = \omega_e W_t + (1 - \omega_e) P_{e,t} \xi$$
(B.16)

$$P_{e,t} e_t^u = \omega_e \,\Delta_t + (1 - \omega_e) \,P_{e,t} \,\xi \tag{B.17}$$

and

$$\frac{P_{e,t} e_t^n}{W_t} = \omega_e + (1 - \omega_e) \frac{P_{e,t} \xi}{W_t}$$
(B.18)

$$\frac{P_{e,t} e_t^u}{\Delta_t} = \omega_e + (1 - \omega_e) \frac{P_{e,t} \xi}{\Delta_t}$$
(B.19)

from this, we see that since $\Delta_t < W_t$, the unemployed spend a higher share of their income on energy consumption.

B.3 Proof of Proposition 2

Since $\delta < w$,

$$\left(\frac{w - p_e \xi}{\delta - p_e \xi}\right) > 1 > \left(\frac{\delta - p_e \xi}{w - p_e \xi}\right)$$

therefore, since both f < 1 and $\lambda = \rho (1 - f) < 1$,

$$(1-\lambda) + \lambda \left(\frac{w - p_e \xi}{\delta - p_e \xi}\right) > 1 > (1-f) + f \left(\frac{\delta - p_e \xi}{w - p_e \xi}\right)$$

implying, in turn, that

$$1 = \beta \left(\frac{1+i}{1+\pi}\right) \left[(1-\lambda) + \lambda \left(\frac{w-p_e\,\xi}{\delta-p_e\,\xi}\right) \right]$$
$$1 > \beta \left(\frac{1+i}{1+\pi}\right) \left[(1-f) + f \left(\frac{\delta-p_e\,\xi}{w-p_e\,\xi}\right) \right]$$

B.4 Constrained-Efficient Allocation

The first-order condition of (37) with respect to n_t gives

$$\ln\left(\frac{w_t^* - \xi \, p_{e,t}}{\delta_t - \xi \, p_{e,t}}\right) + \nu \left[\frac{p_{g,t} - \gamma_e \, p_{e,t}}{1 - \gamma_e} - w_t^* - \left(\frac{1}{1 - \alpha}\right) \, \frac{\kappa}{q_t^*}\right] + \beta \, \mathbb{E}_t \left[\frac{\partial W(n_t, p_{e,t+1})}{\partial n_t}\right] = 0 \quad (B.20)$$

while the envelope condition reads

$$\frac{\partial W(n_{t-1}, p_{e,t})}{\partial n_{t-1}} = \nu \left(\frac{1}{1-\alpha}\right) (1-\rho) \frac{\kappa}{q_t^*} (1-\alpha f_t^*).$$
(B.21)

Combining (B.20) with (B.21) one period ahead and exploiting the fact that $q_t = f_t^{\frac{\alpha}{\alpha-1}}$, one can get the forward recursion for the constrained-efficient job-finding rate of equation (39).

B.5 Second-order Approximation of the Welfare Objective

When $\rho = 1$ and $\alpha = 0.5$,

$$\begin{aligned} U_t &= \ln(\delta_t - \xi \, p_{e,t}) + n_t \left[\ln(w_t - \xi \, p_{e,t}) - \ln(\delta_t - \xi \, p_{e,t}) \right] \\ &+ \nu \left\{ \left[\left(\frac{1}{1 - \gamma_e} \right) \, \frac{p_{g,t}}{\mathcal{D}_t} - \left(\frac{\gamma_e}{1 - \gamma_e} \right) \, p_{e,t} - w_t \right] \, n_t - \kappa \, n_t^2 - \xi \, p_{e,t} \right\} \end{aligned}$$

and, at the constrained-efficient steady state,

$$\frac{\partial U}{\partial n} = \ln(w^* - \xi p_e) - \ln(\delta - \xi p_e) + \nu \left\{ \left[\left(\frac{1}{1 - \gamma_e} \right) p_g - \left(\frac{\gamma_e}{1 - \gamma_e} \right) p_e - w^* \right] - 2\kappa n^* \right\} = 0.$$

In second-order approximation, when $\widetilde{w}_t = -\chi \, \widetilde{p}_{e,t}$,

$$U_t \simeq -\nu \kappa \, \widehat{n}_t^2 + \frac{\partial^2 U}{\partial n \, \partial p_e} \, \widehat{p}_{e,t} \, \widehat{n}_t - \frac{\nu \, n \, p_g}{1 - \gamma_e} \, (\mathcal{D}_t - 1) + \text{t.i.p.}$$

Noticing that, since derivatives are evaluated at the constrained-efficient steady state,

$$\frac{\partial^2 U}{\partial n \, \partial p_e} \, \widehat{p}_{e,t} = 2 \, \kappa \, \nu \, \widehat{n}_t^*$$

and $\hat{n_t} = -\mathcal{U}_t + \hat{n}_t^*$, we have

$$U_t \simeq -\nu \kappa \mathcal{U}_t^2 - \frac{\nu n p_g}{1 - \gamma_e} (\mathcal{D}_t - 1) + \text{t.i.p.}$$

where

$$\mathcal{D}_t - 1 \simeq \ln(\mathcal{D}_t) = \ln \int_0^1 \left[\frac{P_t(k)}{P_{g,t}}\right]^{-\varepsilon} dk \simeq \frac{\varepsilon}{2} \operatorname{var}_k \left\{ \ln(P_t(k)) \right\}$$

Lemma 1.

$$\sum_{t=0}^{\infty} \beta^t \operatorname{var}_k \left\{ \ln(P_t(k)) \right\} = \frac{1}{\Theta} \sum_{t=0}^{\infty} \beta^t \, \pi_{g,t}^2 + t.i.p.$$

Proof. See Chapter 6 in Woodford (2003).

Therefore,

$$\mathbb{E}_t \sum_{j=0}^{\infty} U_{t+j} \simeq -\kappa \,\nu \,\mathbb{E}_t \,\sum_{j=0}^{\infty} \beta^j \,\left(\mathcal{U}_{t+j}^2 + \Omega \,\pi_{g,t+j}^2\right) + \text{t.i.p.}$$
(B.22)

with $\Omega := \frac{n\varepsilon}{(1-\gamma_e)} \frac{p_g}{2\kappa} / \Theta$.

B.6 Optimal Discretionary Policy

The optimality condition of the problem gives

$$\mathcal{U}_t = \left(\frac{n\,\varepsilon}{1-\gamma_e}\right)\,\pi_{g,t}\tag{B.23}$$

which, combined with (56), gives

$$\pi_{g,t} = \frac{\beta}{1 + \Theta \frac{2\kappa}{p_g} \frac{n\varepsilon}{(1-\gamma_e)}} \mathbb{E}_t(\pi_{g,t+1}) + \frac{\Theta \frac{w}{p_g} \zeta \Psi\left(\frac{\Xi_w}{1-\Xi_w}\right) (1+\chi)}{1 + \Theta \frac{2\kappa}{p_g} \frac{n\varepsilon}{(1-\gamma_e)}} \widetilde{p}_{e,t}.$$
 (B.24)

Since $\beta \left(1 + \Theta \frac{2\kappa}{p_g} \frac{n\varepsilon}{(1-\gamma_e)}\right)^{-1} < 1$, the linear expectational difference equation in (B.24) has a unique bounded solution giving, along with the optimality condition in (B.23), the targets for core inflation and the welfare-relevant unemployment gap of equations (58) and (59).

B.7 Optimal Ramsey Policy

The first-order conditions of the problem are, for $t \ge 0$,

$$\Omega \,\pi_{g,t} + \mu_t - \mu_{t-1} = 0 \tag{B.25}$$

$$\mathcal{U}_t + \Theta \, \frac{2 \,\kappa}{p_g} \, \mu_t = 0 \tag{B.26}$$

where μ_t is the Lagrange multiplier associated with the period-*t* constraint and $\mu_{-1} = 0$. Approaching the problem from a *timeless perspective* (see Woodford, 2003), the first-order conditions give

$$\left(\frac{n\,\varepsilon}{1-\gamma_e}\right)\,\pi_{g,t} - \mathcal{U}_t + \mathcal{U}_{t-1} = 0\,. \tag{B.27}$$

_

Combined with the NKPC in (56), this gives

$$\begin{split} \mathcal{U}_{t} &= \frac{\beta}{1+\beta+\Theta\frac{2\kappa}{p_{g}}\frac{n\varepsilon}{(1-\gamma_{e})}} \,\mathbb{E}_{t}(\mathcal{U}_{t+1}) + \frac{1}{1+\beta+\Theta\frac{2\kappa}{p_{g}}\frac{n\varepsilon}{(1-\gamma_{e})}} \,\mathcal{U}_{t-1} \\ &\quad + \frac{\Theta\frac{w}{p_{g}}\frac{n\varepsilon}{(1-\gamma_{e})}}{1+\beta+\Omega\frac{2\kappa}{p_{g}}\frac{n\varepsilon}{(1-\gamma_{e})}} \,\zeta \,\Psi\left(\frac{\Xi_{w}}{1-\Xi_{w}}\right) \,(1+\chi)\,\widetilde{p}_{e,t} \end{split}$$

whose unique stationary solution is

$$\mathcal{U}_{t} = \eta \,\mathcal{U}_{t-1} + \left(\frac{n\,\varepsilon}{1-\gamma_{e}}\right)\,\overline{\Upsilon}\,\zeta\,\Psi\,\left(\frac{\Xi_{w}}{1-\Xi_{w}}\right)\,(1+\chi)\,\widetilde{p}_{e,t} \tag{B.28}$$

where $\eta := \frac{1+\beta+\Theta\frac{2\kappa}{p_g}\frac{n\varepsilon}{(1-\gamma_e)}}{2\beta} \left[1 - \sqrt{1 - 4\beta \left(1 + \beta + \Theta\frac{2\kappa}{p_g}\frac{n\varepsilon}{(1-\gamma_e)}\right)^{-2}}\right] \in (0,1)$ and $\overline{\Upsilon} = \Theta \frac{w}{p_g} \left(1 - \eta \beta + \Theta \frac{2\kappa}{p_g}\frac{n\varepsilon}{(1-\gamma_e)}\right)^{-1}$.

The paths for \mathcal{U}_{t+j} and $\pi_{g,t+j}$ can then be recovered by exploiting (B.27) and (B.28), imposing a steady-state initial condition (i.e. $\mathcal{U}_{t-1} = 0$) and considering a one-off shock at t with persistence $\rho_e \to 1$.

References

- Acharya, Sushant, Edouard Challe, and Keshav Dogra. 2023. "Optimal monetary policy according to HANK." American Economic Review, 113(7): 1741–1782.
- Ari, Anil, Nicolas Arregui, Simon Black, Oya Celasun, Dora M. Iakova, Aiko Mineshima, Victor Mylonas, Ian W.H. Parry, Iulia Teodoru, and Karlygash Zhunussova. 2022. "Surging Energy Prices in Europe in the Aftermath of the War: How to Support the Vulnerable and Speed up the Transition Away from Fossil Fuels." IMF Working Paper 152.
- Arregui, Nicolas, Oya Celasun, Dora M. Iakova, Aiko Mineshima, Victor Mylonas, Frederik G. Toscani, Yu Ching Wong, Li Zeng, and Jing Zhou. 2022. "Targeted, Implementable, and Practical Energy Relief Measures for Households in Europe." IMF Working Paper 262.
- Bachmann, Ruediger, David Baqaee, Christian Bayer, Moritz Kuhn, Andreas Löschel, Benjamin Moll, Andreas Peichl, Karen Pittel, and Moritz Schularick. 2022. "What if? The economic effects for Germany of a stop of energy imports from Russia." CESifo EconPol Policy Report 36.
- Battistini, Niccolò, Virginia Di Nino, Maarten Dossche, and Aleksandra Kolndrekaj. 2022. "Energy prices and private consumption: what are the channels?" European Central Bank Economic Bulletin 3.
- Blanchard, Olivier J., and Jordi Galí. 2007. "The Macroeconomic Effects of Oil Shocks: Why are the 2000s so different from the 1970s?" NBER Working Paper 13368.
- Blanchard, Olivier J., and Jordi Galí. 2010. "Labor markets and monetary policy: A New Keynesian model with unemployment." *American Economic Journal: Macroeconomics*, 2(2): 1–30.
- Blanchard, Olivier J., and Marianna Riggi. 2013. "Why are the 2000s so different from the 1970s? A structural interpretation of changes in the macroeconomic effects of oil prices." *Journal of the European Economic Association*, 11(5): 1032–1052.
- Bodenstein, Martin, Christopher J. Erceg, and Luca Guerrieri. 2008. "Optimal monetary policy with distinct core and headline inflation rates." *Journal of Monetary Economics*, 55: S18–S33.

- Challe, Edouard. 2020. "Uninsured unemployment risk and optimal monetary policy in a zeroliquidity economy." *American Economic Journal: Macroeconomics*, 12(2): 241–83.
- Challe, Edouard, Julien Matheron, Xavier Ragot, and Juan F. Rubio-Ramirez. 2017. "Precautionary saving and aggregate demand." *Quantitative Economics*, 8(2): 435–478.
- Chan, Jenny, Sebastian Diz, and Derrick Kanngiesser. 2023. "Energy Prices and Household Heterogeneity: Monetary Policy in a Gas-TANK." Bank of England Staff Working Paper 1041.
- Chodorow-Reich, Gabriel, and Loukas Karabarbounis. 2016. "The cyclicality of the opportunity cost of employment." *Journal of Political Economy*, 124(6): 1563–1618.
- **Corsello, Francesco, and Marianna Riggi.** 2023. "Inflation is not equal for all: disparities in the macroeconomic effects of energy shocks." Bank of Italy Working Paper 1429.
- Curci, Nicola, Marco Savegnago, Giordano Zevi, and Roberta Zizza. 2022. "The redistributive effects of inflation: a microsimulation analysis for Italy." Bank of Italy Occasional Paper 738.
- **Debortoli, Davide, and Jordi Galí.** 2017. "Monetary policy with heterogeneous agents: Insights from TANK models." Universitat Pompeu Fabra Economics Working Paper 1686.
- Den Haan, Wouter J., Pontus Rendahl, and Markus Riegler. 2018. "Unemployment (fears) and deflationary spirals." *Journal of the European Economic Association*, 16(5): 1281–1349.
- Gagliardone, Luca, and Mark Gertler. 2023. "Oil Prices, Monetary Policy, and Inflation Surges." NBER Working Paper 31263.
- Känzig, Diego R. 2023. "The unequal economic consequences of carbon pricing." NBER Working Paper 31221.
- Kaplan, Greg, Benjamin Moll, and Giovanni L. Violante. 2018. "Monetary policy according to HANK." American Economic Review, 108(3): 697–743.
- Kharroubi, Enisse, and Frank Smets. 2023. "Energy shocks as Keynesian supply shocks: implications for fiscal policy." BIS Working Paper 1151.
- Kouvavas, Omiros, Chiara Osbat, Timo Reinelt, and Isabel Vansteenkiste. 2021. "Markups and inflation cyclicality in the euro area." ECB Working Paper 2617.

- McKay, Alisdair, and Ricardo Reis. 2021. "Optimal automatic stabilizers." The Review of Economic Studies, 88(5): 2375–2406.
- Montoro, Carlos. 2012. "Oil shocks and optimal monetary policy." *Macroeconomic Dynamics*, 16(2): 240–277.
- Mortensen, Dale T., and Christopher A. Pissarides. 1994. "Job creation and job destruction in the theory of unemployment." *The Review of Economic Studies*, 61(3): 397–415.
- Natal, Jean-Marc. 2012. "Monetary policy response to oil price shocks." Journal of Money, Credit and Banking, 44(1): 53–101.
- **Petrongolo, Barbara, and Christopher A. Pissarides.** 2001. "Looking into the black box: A survey of the matching function." *Journal of Economic Literature*, 39(2): 390–431.
- Pieroni, Valerio. 2023. "Energy shortages and aggregate demand: Output loss and unequal burden from HANK." *European Economic Review*, 154: 104428.
- Ravn, Morten O., and Vincent Sterk. 2017. "Job uncertainty and deep recessions." *Journal* of Monetary Economics, 90: 125–141.
- Ravn, Morten O., and Vincent Sterk. 2021. "Macroeconomic fluctuations with HANK & SAM: An analytical approach." *Journal of the European Economic Association*, 19(2): 1162–1202.
- Smirnov, Danila. 2022. "Optimal Monetary Policy in HANK." Unpublished.
- Woodford, Michael. 2003. Interest and Prices: Foundations of a Theory of Monetary Policy. Princeton, NJ: Princeton University Press.

RECENTLY PUBLISHED "TEMI" (*)

- N. 1419 *Temperatures and search: evidence from the housing market*, by Michele Cascarano and Filippo Natoli (July 2023).
- N. 1420 Flight to climatic safety: local natural disasters and global portfolio flows, by Fabrizio Ferriani, Andrea Gazzani and Filippo Natoli (July 2023).
- N. 1421 The effects of the pandemic on households' financial savings: a Bayesian structural VAR analysis, by Luigi Infante, Francesca Lilla and Francesco Vercelli (October 2023).
- N. 1422 *Decomposing the monetary policy multiplier*, by Piergiorgio Alessandri, Fabrizio Venditti and Oscar Jordà (October 2023).
- N. 1423 The short and medium term effects of full-day schooling on learning and maternal labor supply, by Giulia Bovini, Nicolò Cattadori, Marta De Philippis and Paolo Sestito (October 2023).
- N. 1424 Subsidizing business entry in competitive credit markets, by Vincenzo Cuciniello, Claudio Michelacci and Luigi Paciello (October 2023).
- N. 1425 Drivers of large recessions and monetary policy responses, by Giovanni Melina and Stefania Villa (October 2023).
- N. 1426 *The performance of household-held mutual funds: evidence from the euro area*, by Valerio Della Corte and Raffaele Santioni (November 2023).
- N. 1427 Trade in the time of COVID-19: an empirical analysis based on Italian data, by Gianmarco Cariola (November 2023).
- N. 1428 Natural gas and the macroeconomy: not all energy shocks are alike, by Piergiorgio Alessandri and Andrea Giovanni Gazzani (November 2023).
- N. 1429 Inflation is not equal for all: the heterogenous effects of energy shocks, by Francesco Corsello and Marianna Riggi (November 2023).
- N. 1430 *Labor market dynamics and geographical reallocations*, by Gaetano Basso, Salvatore Lo Bello and Francesca Subioli (November 2023).
- N. 1431 Monetary and fiscal policy responses to fossil fuel price shocks, by Anna Bartocci, Alessandro Cantelmo, Pietro Cova, Alessandro Notarpietro and Massimiliano Pisani (December 2023).
- N. 1432 Do female leaders choose women? Evidence from visible and hidden appointments, by Andrea Cintolesi and Edoardo Frattola (December 2023).
- N. 1433 Monetary policy tightening in response to uncertain stagflationary shocks: a modelbased analysis, by Anna Bartocci, Alessandro Cantelmo, Alessandro Notarpietro and Massimiliano Pisani (December 2023).
- N. 1434 *Inflation, capital structure and firm value*, by Andrea Fabiani and Fabio Massimo Piersanti (December 2023).
- N. 1435 Announcement and implementation effects of central bank asset purchases, by Marco Bernardini and Antonio M. Conti (December 2023).
- N. 1436 *Connecting the dots: the network nature of shocks propagation in credit markets*, by Stefano Pietrosanti and Edoardo Rainone (December 2023).
- N. 1437 Inflation expectations and misallocation of resources: evidence from Italy, by Tiziano Ropele, Yuriy Gorodnichenko and Olivier Coibion (December 2023).
- N. 1438 Women in economics: the role of gendered references at entry in the profession, by Audinga Baltrunaite, Alessandra Casarico and Lucia Rizzica (February 2024).
- N. 1439 *Procuring survival*, by Matilde Cappelletti, Leonardo M. Giuffrida and Gabriele Rovigatti (February 2024).
- N. 1440 Estimating the returns to occupational licensing: evidence from regression discontinuities at the bar exam, by Omar Bamieh, Andrea Cintolesi and Mario Pagliero (February 2024).
- N. 1441 Household perceived sources of business cycle fluctuations: a tale of supply and demand, by Clodomiro Ferreira and Stefano Pica (February 2024).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

2022

- ANDINI M., M. BOLDRINI, E. CIANI, G. DE BLASIO, A. D'IGNAZIO and A. PALADINI, Machine learning in the service of policy targeting: the case of public credit guarantees, Journal of Economic Behavior & Organization, v. 198, pp. 434-475, WP 1206 (February 2019).
- ANGELICO C., J. MARCUCCI, M. MICCOLI and F. QUARTA, Can we measure inflation expectations using twitter?, Journal of Econometrics, v. 228, 2, pp. 259-277, WP 1318 (February 2021).
- BARTOCCI A., A. NOTARPIETRO and M. PISANI, *Covid-19 shock and fiscal-monetary policy mix in a monetary union*, Economic challenges for Europe after the pandemic, Springer Proceedings in Business and Economics, Berlin-Heidelberg, Springer, **WP 1313 (December 2020).**
- BOTTERO M., C. MINOIU, J. PEYDRÒ, A. POLO, A. PRESBITERO and E. SETTE, *Expansionary yet different: credit supply and real effects of negative interest rate policy*, Journal of Financial Economics, v. 146, 2, pp. 754-778, WP 1269 (March 2020).
- BRONZINI R., A. D'IGNAZIO and D. REVELLI, *Financial structure and bank relationships of Italian multinational firms*, Journal of Multinational Financial Management, v. 66, Article 100762, WP 1326 (March 2021).
- CANTELMO A., *Rare disasters, the natural interest rate and monetary policy,* Oxford Bulletin of Economics and Statistics, v. 84, 3, pp. 473-496, **WP 1309 (December 2020).**
- CARRIERO A., F. CORSELLO and M. MARCELLINO, *The global component of inflation volatility,* Journal of Applied Econometrics, v. 37, 4, pp. 700-721, WP 1170 (May 2018).
- CIAPANNA E. and G. ROVIGATTI, *The grocery trolley race in times of Covid-19. Evidence from Italy*, Italian Economic Journal / Rivista italiana degli economisti, v. 8, 2, pp. 471-498, **WP 1341 (June 2021)**.
- CONTI A. M., A. NOBILI and F. M. SIGNORETTI, *Bank capital requirement shocks: a narrative perspective,* European Economic Review, v.151, Article 104254, **WP 1199 (November 2018).**
- FAIELLA I. and A. MISTRETTA, *The net zero challenge for firms' competitiveness*, Environmental and Resource Economics, v. 83, pp. 85-113, **WP 1259 (February 2020).**
- FERRIANI F. and G. VERONESE, *Hedging and investment trade-offs in the U.S. oil industry*, Energy Economics, v. 106, Article 105736, WP 1211 (March 2019).
- GUISO L., A. POZZI, A. TSOY, L. GAMBACORTA and P. E. MISTRULLI, *The cost of steering in financial markets:* evidence from the mortgage market, Journal of Financial Economics, v.143, 3, pp. 1209-1226, WP 1252 (December 2019).
- LAMORGESE A. and D. PELLEGRINO, *Loss aversion in housing appraisal: evidence from Italian homeowners,* Journal of Housing Economics, v. 56, Article 101826, WP 1248 (November 2019).
- LI F., T. MÄKINEN, A. MERCATANTI and A. SILVESTRINI, *Causal analysis of central bank holdings of corporate bonds under interference*, Economic Modelling, v.113, Article 105873, WP 1300 (November 2020).
- LOBERTO M, A. LUCIANI and M. PANGALLO, *What do online listings tell us about the housing market?*, International Journal of Central Banking, v. 18, 4, pp. 325-377, **WP 1171 (April 2018).**
- MIRENDA L., M. SAURO and L. RIZZICA, *The economic effects of mafia: firm level evidence*, American Economic Review, vol. 112, 8, pp. 2748-2773, WP 1235 (October 2019).
- MOCETTI S., G. ROMA and E. RUBOLINO, *Knocking on parents' doors: regulation and intergenerational mobility*, Journal of Human Resources, v. 57, 2, pp. 525-554, WP 1182 (July 2018).
- PERICOLI M. and M. TABOGA, Nearly exact Bayesian estimation of non-linear no-arbitrage term-structure models, Journal of Financial Econometrics, v. 20, 5, pp. 807-838, WP 1189 (September 2018).
- ROSSI P. and D. SCALISE, *Financial development and growth in European regions,* Journal of Regional Science, v. 62, 2, pp. 389-411, WP 1246 (November 2019).
- SCHIVARDI F., E. SETTE and G. TABELLINI, *Credit misallocation during the European financial crisis,* Economic Journal, v. 132, 641, pp. 391-423, **WP 1139 (September 2017).**
- TABOGA M., Cross-country differences in the size of venture capital financing rounds: a machine learning approach, Empirical Economics, v. 62, 3, pp. 991-1012, WP 1243 (November 2019).

2023

- APRIGLIANO V., S. EMILIOZZI, G. GUAITOLI, A. LUCIANI, J. MARCUCCI and L. MONTEFORTE, *The power of text-based indicators in forecasting Italian economic activity*, International Journal of Forecasting, v. 39, 2, pp. 791-808, WP 1321 (March 2021).
- BARTOCCI A., A. NOTARPIETRO and M. PISANI, *Non-standard monetary policy measures in non-normal times,* International Finance, v. 26, 1, pp. 19-35, **WP 1251 (November 2019).**

- CAPPELLETTI G. and P. E. MISTRULLI, *The role of credit lines and multiple lending in financial contagion and systemic events*, Journal of Financial Stability, v. 67, Article 101141, WP 1123 (June 2017).
- CECI D. and A. SILVESTRINI, *Nowcasting the state of the Italian economy: the role of financial markets,* Journal of Forecasting, v. 42, 7, pp. 1569-1593, **WP 1362 (February 2022).**
- CIAPANNA E, S. MOCETTI and A. NOTARPIETRO, *The macroeconomic effects of structural reforms: an empirical and model-based approach*, Economic Policy, v. 38, 114, pp. 243-285, **WP 1303 (November 2020).**
- DAURICH D, S. DI ADDARIO and R. SAGGIO, The macroeconomic effects of structural reforms: an empirical and model-based approach, Review of Economic Studies, v. 90, 6, pp. 2880–2942, WP 1390 (November 2022).
- DI ADDARIO S., P. KLINE, R. SAGGIO and M. SØLVSTEN, *The effects of partial employment protection reforms:* evidence from Italy, Journal of Econometrics, v. 233, 2, pp. 340-374, WP 1374 (June 2022).
- FERRARI A. and V. NISPI LANDI, *Toward a green economy: the role of central bank's asset purchases,* International Journal of Central Banking, v. 19, 5, pp. 287-340, WP 1358 (February 2022).
- FERRIANI F., *Issuing bonds during the Covid-19 pandemic: was there an ESG premium?*, International Review of Financial Analysis, v. 88, Article 102653, **WP 1392 (November 2022).**
- GIORDANO C., Revisiting the real exchange rate misalignment-economic growth nexus via the across-sector misallocation channel, Review of International Economics, v. 31, 4, pp. 1329-1384, WP 1385 (October 2022).
- GUGLIELMINETTI E., M. LOBERTO and A. MISTRETTA, *The impact of COVID-19 on the European short-term rental market*, Empirica, v. 50, 3, pp. 585-623, **WP 1379 (July 2022).**
- LILLA F., Volatility bursts: a discrete-time option model with multiple volatility components, Journal of Financial Econometrics, v. 21, 3, pp. 678-713, WP 1336 (June 2021).
- LOBERTO M., *Foreclosures and house prices*, Italian Economic Journal / Rivista italiana degli economisti, v. 9, 1, pp. 397-424, **WP 1325 (March 2021).**
- LOMBARDI M. J., M. RIGGI and E. VIVIANO, Worker's bargaining power and the Phillips curve: a micro-macro analysis, and wages, Journal of the European Economic Association, v. 21, 5, pp. 1905–1943, WP 1302 (November 2020).
- NERI S., Long-term inflation expectations and monetary policy in the Euro Area before the pandemic, European Economic Review, v. 154, Article 104426, WP 1357 (December 2021).
- ORAME A., *Bank lending and the European debt crisis: evidence from a new survey*, International Journal of Central Banking, v. 19, 1, pp. 243-300, **WP 1279 (June 2020).**
- RIZZICA L., G. ROMA and G. ROVIGATTI, *The effects of shop opening hours deregulation: evidence from Italy,* The Journal of Law and Economics, v. 66, 1, pp. 21-52, **WP 1281 (June 2020).**
- TANZI G. M., Scars of youth non-employment and labour market conditions, Italian Economic Journal / Rivista italiana degli economisti, v. 9, 2, pp. 475-499, WP 1312 (December 2020).

2024

MORO A. and V. NISPI LANDI, *The external financial spillovers of CBDCs*, Journal of Economic Dynamics and Control, v. 159, Article 104801, **WP 1416 (July 2023).**

FORTHCOMING

- BALTRUNAITE A., M. CANNELLA, S. MOCETTI and G. ROMA, Board composition and performance of state-owned enterprises: quasi experimental evidence, The Journal of Law, Economics, and Organization, WP 1328 (April 2021).
- BUONO I, F. CORNELI and E. DI STEFANO, *Capital inflows to emerging countries and their sensitivity to the global financial cycle*, International Finance, WP 1262 (February 2020).
- CORNELI F., Sovereign debt maturity structure and its costs, International Tax and Public Finance, WP 1196 (November 2018).
- CUCINIELLO V. and N. DI IASIO, Determinants of the credit cycle: a flow analysis of the extensive margin, Journal of Money, Credit and Banking, WP 1266 (March 2020).
- FERRARI A. and V. NISPI LANDI, Whatever it takes to save the planet? Central banks and unconventional green policy, Macroeconomic Dynamics, WP 1320 (February 2021).
- FLACCADORO M., *Exchange rate pass-through in small, open, commodity-exporting economies: lessons from Canada,* Journal of International Economics, **WP 1365 (April 2022).**

- GAUTIER E., C. CONFLITTI, R. FABER, B. FABO, L. FADEJEVA, V. JOUVANCEAU, J.-O. MENZ, T. MESSNER, P. PETROULAS, P. ROLDAN-BLANCO, F. RUMLER, S. SANTORO, E. WIELAND and H. ZIMMER, *New facts on consumer price rigidity in the euro area,* American Economic Journal: Macroeconomics, **WP 1375** (July 2022).
- MICHELANGELI V. and E. VIVIANO, Can internet banking affect households' participation in financial markets and financial awarness?, Journal of Money, Credit and Banking, WP 1329 (April 2021).
- MISTRETTA A., Synchronization vs transmission: the effect of the German slowdown on the Italian business cycle, International Journal of Central Banking, WP 1346 (October 2021).
- RAINONE E., *Reservation rates in interbank money markets,* Journal of Money, Credit and Banking, WP 1160 (February 2021).
- RAINONE E., *Real-time identification and high frequency analysis of deposits outflows*, Journal of Financial Econometrics, **WP 1319 (December 2017).**